

\documentclass[a4paper,11pt]{amsart}

\newfont{\cyr}{wncyr10 scaled 1100}

\setcounter{tocdepth}{1}


\usepackage[left=2.7cm,right=2.7cm,top=3.5cm,bottom=3cm]{geometry}
\usepackage{amsthm,amssymb,amsmath,amsfonts,mathrsfs,amscd}
\usepackage[latin1]{inputenc}
\usepackage[all]{xy}
\usepackage{latexsym}
\usepackage{longtable}








 
\theoremstyle{plain}
\newtheorem{theorem}{Theorem}[section]
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{conjecture}[theorem]{Conjecture}

\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{question}[theorem]{Question}
\newtheorem{examplewr}[theorem]{Example}
\newtheorem{ass}[theorem]{Assumption}
\newtheorem{app}[theorem]{Application}

\theoremstyle{remark}
\newtheorem{obswr}[theorem]{Observation}
\newtheorem{remarkwr}[theorem]{Remark}

\newenvironment{obs}{\begin{obswr}\begin{upshape}}{\end{upshape}\end{obswr}}
\newenvironment{remark}{\begin{remarkwr}\begin{upshape}}{\end{upshape}\end{remarkwr}}
\newenvironment{example}{\begin{examplewr}\begin{upshape}}{\end{upshape}\end{examplewr}}

\DeclareMathOperator{\fil}{fil}
\DeclareMathOperator{\fr}{fr}


\DeclareMathOperator{\an}{an}
\DeclareMathOperator{\nh}{nh}
\DeclareMathOperator{\cusps}{cusps}
\DeclareMathOperator{\ffm}{ffm}
\DeclareMathOperator{\ppar}{par}
\DeclareMathOperator{\alg}{alg}





\DeclareMathOperator{\lcm}{lcm}
\DeclareMathOperator{\dR}{dR}
\DeclareMathOperator{\fp}{fp}
\DeclareMathOperator{\cris}{cris}
\DeclareMathOperator{\Frac}{Frac}
\DeclareMathOperator{\et}{et}
\DeclareMathOperator{\syn}{syn}
\DeclareMathOperator{\oc}{oc}
\DeclareMathOperator{\ur}{u-r}
\DeclareMathOperator{\ah}{a-h}
\DeclareMathOperator{\para}{par}

\DeclareMathOperator{\cts}{cts}
\DeclareMathOperator{\bal}{bal}
\DeclareMathOperator{\KS}{KS}
\DeclareMathOperator{\sym}{\mathrm{sym}}
\DeclareMathOperator{\Sym}{Sym}
\DeclareMathOperator{\Ad}{Ad}
\DeclareMathOperator{\sgn}{sgn}
\DeclareMathOperator{\spec}{spec}
\DeclareMathOperator{\la}{la} 
\DeclareMathOperator{\Ind}{Ind}
\DeclareMathOperator{\ad}{Ad}

\newcommand{\GoneN}{\Gamma_1(N)}
\newcommand{\XoneN}{X_1(N)}
\newcommand{\Mk}{{\mathcal M}_k(\Gamma_1(N))}
\newcommand{\Sk}{{\mathcal S}_k(\Gamma_1(N))}

\newcommand{\Res}{{\rm Res}}

\newcommand{\ka}{{\kappa}}


\newcommand{\uomega}{{\underline{\omega}}}
\newcommand{\cZ}{{\mathcal Z}}
\newcommand{\hf}{{\mathbf{f}}}
\newcommand{\hg}{{\mathbf{g}}}
\newcommand{\hh}{{\mathbf {h}}}
\newcommand{\hE}{{\mathbf {E}}}
% For below, try \breve, \check, \dot, \hat, \mathring, \ddot
\newcommand{\testf}{\breve{f}}
\newcommand{\testg}{\breve{g}} 
\newcommand{\testh}{\breve{h}}
\newcommand{\testgamma}{\breve{\gamma}}
\newcommand{\testG}{\breve{G}}
\newcommand{\testH}{\breve{H}}
\newcommand{\htf}{\breve{\mathbf{f}}}
\newcommand{\htg}{\breve{\mathbf{g}}}
\newcommand{\hth}{\breve{\mathbf {h}}}
\newcommand{\uS}{{\mathbf{S}}}
\newcommand{\hphi}{\boldsymbol{\phi}}
\newcommand{\uhp}{{\mathfrak{H}}}
\newcommand{\PPhi}{{\underline{\phi}}}
\newcommand{\ff}{{\mathbf{f}}}
\newcommand{\aaa}{{\mathbf{a}}}
\newcommand{\LL}{{\mathscr{L}}}
\newcommand{\Lp}{{\mathscr{L}_p}}
\newcommand{\varrhop}{\varrho(P,\omega_2,\omega_3)}
\newcommand{\ukappa}{\underline{\kappa}}
\newcommand{\uV}{{\bf V}}

\newcommand{\fa}{{\mathfrak{a}}}
\newcommand{\fb}{{\mathfrak{b}}}
\newcommand{\fc}{{\mathfrak{c}}}
\newcommand{\fd}{{\mathfrak{d}}}


\newcommand{\hB}{{\mathbf B}}
\newcommand{\St}{{(\infty)}}
\newcommand{\cl}{{\mathrm{cl}}}
\DeclareMathOperator{\red}{red}

\newcommand{\cA}{\mathcal A}
\newcommand{\cN}{\mathcal N}
\newcommand{\cW}{\mathcal W}
\newcommand{\cV}{\mathcal V}
\newcommand{\cF}{\mathcal F}
\newcommand{\cX}{\mathcal X}
\newcommand{\cH}{\mathcal H}
\newcommand{\cC}{\mathcal C}
\newcommand{\V}{\mathcal V}
\newcommand{\E}{\mathcal E}
\newcommand{\cE}{\mathcal E}
\newcommand{\g}{\gamma}
% \newcommand{\G}{\Gamma}
\newcommand{\Gap}{\Gamma_0^D(p M)}
\newcommand{\Ga}{\Gamma_0^D(M)}
\newcommand{\hGa}{\hat{\Gamma }_0(M)}
\newcommand{\Q}{\mathbb{Q}}
\newcommand{\A}{\mathbb{A}}
\newcommand{\Z}{\mathbb{Z}}
\newcommand{\G}{\mathbb{G}}
\newcommand{\F}{\mathbb{F}}
\newcommand{\K}{\mathbb{K}}
\newcommand{\C}{\mathbb{C}}
\newcommand{\I}{\mathbb{I}}
\newcommand{\Ql}{\Q_{\ell}}
\newcommand{\GQ}{G_{\Q}}
\newcommand{\PP}{\mathbb{P}}
\newcommand{\Sel}{\mathrm{Sel}}
\newcommand{\Sha}{\mathrm{Sha}}
\newcommand{\Gal}{\mathrm{Gal\,}}
\newcommand{\GL}{\mathrm{GL}}
\newcommand{\Cl}{\mathrm{Cl}}
\newcommand{\Tr}{\mathrm{Tr}}
\newcommand{\Div}{\mathrm{Div}}
\newcommand{\Fil}{\mathrm{Fil}}
\newcommand{\cond}{\mathrm{cond}}
\newcommand{\Frob}{\mathrm{Frob}}
\newcommand{\End}{\mathrm{End}}
\newcommand{\Jac}{\mathrm{Jac}}
\newcommand{\disc}{\mathrm{disc}}
\newcommand{\Id}{\mathrm{Id}}
\newcommand{\MS}{\mathrm{MS}}
\newcommand{\Aut}{\mathrm{Aut}}
\newcommand{\Fr}{\mathrm{Fr}}
\newcommand{\diag}{\mathrm{diag}}
\newcommand{\Up}{\mathrm{U}_p}
\newcommand{\ord}{{\mathrm{ord}}}
\newfont{\gotip}{eufb10 at 12pt}
\newcommand{\gn}{\mbox{\gotip n}}
\newcommand{\gm}{\mbox{\gotip m}}
\newcommand{\ga}{\mbox{\gotip a}}
\newcommand{\gb}{\mbox{\gotip b}}
\newcommand{\gf}{\mbox{\gotip f}}
\newcommand{\go}{{\mathcal O}_{A_f}}
\newcommand{\OAf}{{\mathcal O}_{A_f}}
\newcommand{\gp}{\mbox{\gotip p}}
\newcommand{\gL}{\mbox{\gotip L}}
\newcommand{\gd}{\mbox{\gotip d}}
\newcommand{\cO}{{\mathcal O}}
\newcommand{\cM}{{\mathcal M}}
\newcommand{\cP}{{\mathcal P}}
\newcommand{\cL}{{\mathcal L}}
\newcommand{\cG}{{\mathcal G}}
\newcommand{\GG}{{\mathcal G}}
\newcommand{\cD}{{\mathcal D}}
\newcommand{\ra}{\rightarrow}
\newcommand{\lra}{\longrightarrow}
\newcommand{\hra}{\hookrightarrow}
\newcommand{\qbar}{\bar\Q}
\newcommand{\Norm}{\mathrm{N}}
\newcommand{\NS}{\mathrm{NS}}
\newcommand{\SL}{{\mathrm {SL}}}
\newcommand{\SO}{{\mathrm {SO}}}
\newcommand{\n}{{\mathrm{n}}}
\newcommand{\Pic}{{\mathrm{Pic}}}
\newcommand{\CH}{{\mathrm{CH}}}
\newcommand{\AJ}{{\mathrm{AJ}}}
\newcommand{\tr}{{\mathrm{tr}}}
\newcommand{\si}{\sigma}
\newcommand{\eps}{\epsilon}
\newcommand{\ta}{{\tau}}
\newcommand{\R}{{\mathbb R}}
\newcommand{\M}{{\mathrm{M}}}
\newcommand{\HC}{\mathcal F_{\rm har}}
\newcommand{\PGL}{{\mathrm{PGL}}}
\newcommand{\Supp}{\mathrm{Supp}}
\newcommand{\ep}{\varepsilon}

\def \mint {\times \hskip -1.1em \int}

%\newcommand{\mint}{\times\!\!\!\!\!\hskip-0.3em\int}

\newcommand{\SP}{\vspace{0.5cm}}
\newcommand{\T}{\mathbb T}
\newcommand{\new}{\mathrm{new}}
\DeclareMathOperator{\Hom}{Hom} 
\DeclareMathOperator{\Meas}{Meas}
\DeclareMathOperator{\cyc}{cyc}
\newcommand{\X}{\mathbb X}
\newcommand{\Y}{\mathbb Y}
\newcommand{\U}{\mathbb U}
\newcommand{\Rat}{{\rm Rat}\bigl(\PP^1(\Q_p),\C_p\bigr)}
\newcommand{\Ker}{{\mathrm Ker}}
\newcommand{\res}{\mathrm{res}}
\DeclareMathOperator{\rig}{rig}
\DeclareMathOperator{\cmp}{comp}
\DeclareMathOperator{\can}{can}
\DeclareMathOperator{\fin}{fin}
\DeclareMathOperator{\even}{even}
\DeclareMathOperator{\spl}{Spl}
\DeclareMathOperator{\hdg}{hdg}
 \DeclareMathOperator{\noc}{n-oc}
 \DeclareMathOperator{\hol}{hol}
 \newcommand{\Corr}{\mathrm{Corr}}
 \DeclareMathOperator{\frob}{Frob}

\newcommand{\TT}{\mathbb{T}}
\newcommand{\uTT}{\underline{\mathbb{T}}}
\newcommand{\D}{\mathbb D}


\newcommand{\rom}{\mathrm}
 

\newcommand{\longmono}{\mbox{$\lhook\joinrel\longrightarrow$}}
\newcommand{\longleftmono}{\mbox{$\longleftarrow\joinrel\varrhook$}}
\newcommand{\longepi}{\mbox{$\relbar\joinrel\twoheadrightarrow$}}
\newcommand{\mat}[4]{\left(\begin{array}{cc}#1&#2\\#3&#4\end{array}\right)}
\newcommand{\smallmat}[4]{\bigl(\begin{smallmatrix}#1&#2\\#3&#4\end{smallmatrix}\bigr)}
\newcommand{\dirlim}{\mathop{\varinjlim}\limits}
\newcommand{\invlim}{\mathop{\varprojlim}\limits}

\def\thetak{\theta^{^{(2k)}}}
\def\thetam{\theta^{^{(2m)}}}
\def\thetakk{\theta^{^{(2k+2)}}}

\def\thetakv{\theta^{^{(2k)\vee}}}
\def\thetamv{\theta^{^{(2m)\vee}}}
\def\thetakkv{\theta^{^{(2k+2)\vee}}}



\def\bN{{\mathbf N}}
\def\Gammatau{{\Gamma_{\!\tau}}}
\def\tkappa{{\tilde\kappa}}
\def\ukappa{{\underline\kappa}}
\def\utheta{{\mbox{\boldmath$\theta$}}}
\def\trho{{\tilde\varrho}}
\def\lra{{\longrightarrow}}
\def\SL{{\bf SL}}
\def\GL{{\bf GL}}
\def\PGL{{\bf PGL}}
\def\PSL{{\bf PSL}}
\def\PP{{\mathbf P}}
\def\X{\mathbf{X}}
\def\A{\mathbf{A}}
\def\rp1{r\!\!+\!\!1}
\def\Lp{{\mathscr L}_p}
\newcommand{\om}{{\omega}}
\def\bp{{x}}
\def\bpp{{y}}


%FUDGE FACTORS
\newcommand{\fhr}{{\mathfrak f_{\mathrm{HR}}}}
\newcommand{\fbdp}{{\mathfrak f_{\mathrm{BDP}}}}
\newcommand{\fkatz}{{\mathfrak f_{K}}}
\newcommand{\fpet}{{\mathfrak f_{\mathrm{Pet}}}}
%EULER FACTORS
\newcommand{\ehr}{{\mathfrak e_{\mathrm{HR}}}}
\newcommand{\ebdp}{{\mathfrak e_{\mathrm{BDP}}}}
\newcommand{\ekatz}{{\mathfrak e_{K}}}

\newcommand{\eul}{{\matcal \varepsilon ul}}


\newcommand{\cEul}{\mathcal Eul}

\newcommand{\CC}{{\mathcal C}}
\newcommand{\UU}{{\mathcal U}}


\newcommand{\chiNN}{{\chi_{_N}}}



\newcommand{\wq}{{\mathfrak q}}
\newcommand{\wqbar}{{\overline{\mathfrak q}}}







\include{thebibliography}

\begin{document}
\begin{thebibliography}{99}

 

%\bibitem[AL]{AL}
%A. O. L. Atkin, Wein-Ch'ing Winnie Li, {\em
%Twists of new forms and pseudo-eigenvalues of $W$-operators}
%Inventiones mathematic, {\bf 48} (1978), Issue 3, 221-243

% \bibitem[BeDa]{BelDas}  J. Bella\"iche, S. Dasgupta,  {\em The $p$-adic $L$-functions of evil Eisenstein series, available at

%\bibitem[BeDa1]{BD-genus}
%M. Bertolini, H. Darmon, {\em  
%The rationality of Stark-Heegner points over genus fields of real quadratic fields }
%Annals of Mathematics, {\bf 170} (2009) 343-369. 

\bibitem[BeDa2]{BD1}
M. Bertolini, H. Darmon, {\em Kato's Euler system and rational points on elliptic curves I: A p-adic Beilinson formula}. 
Israel Journal of Mathematics, {\bf 199}, Issue 1, January 2014, 163-188.

\bibitem[BeDi]{BeDi} J. Bella\"iche, M. Dimitrov, {\em On the eigencurve at classical weight one points}, to appear in Duke Math. J., available at {\tt http://math.univ-lille1.fr/$\sim$mladen/}


\bibitem[BDP1]{BDP1}
M. Bertolini, H. Darmon, K. Prasanna, {\em  Generalised Heegner cycles and $p$-adic
Rankin L-series}, Duke Math J. {\bf 162} No. 6,  (2013) pp. 1033-1148.

\bibitem[BDP2]{BDP2}
M. Bertolini, H. Darmon, K. Prasanna, {\em  p-adic Rankin L-series and rational points on CM elliptic curves}, 
Pacific Journal of Mathematics, Vol. 260, No. 2 (2012), pp. 261-303.


\bibitem[BDR1]{BDR1}
M.~Bertolini, H.~Darmon and V.~Rotger, 
{\em Beilinson-Flach elements and Euler Systems I: syntomic regulators and $p$-adic Rankin L-series}, Journal of Algebraic Geometry, {\bf 24} (2015), 355-378

\bibitem[BDR2]{BDR2} 
M.\ Bertolini, H.\ Darmon, V.\ Rotger, {\em
Beilinson-Flach elements and Euler systems II: the Birch and Swinnerton-Dyer conjecture for Hasse-Weil-Artin L-series }, Journal of Algebraic Geometry, {\bf 24} (2015), 569-604.



%\bibitem[BDST]{bdst}
%K.~Buzzard, M.~Dickinson, N.~Shepherd-Barron, R.~Taylor.
%{\em  On icosahedral Artin representations}, 
%Duke Math. J.  {\bf 109}  (2001),  no. 2, 283--318. 

% \bibitem[Ci]{Ci} C. Citro, IMRN.

% \bibitem[CS]{CoSc} J. Coates, C.-G. Schmidt, {\em Iwasawa theory for the symmetric square of an elliptic curve}, J. reine angew. Math. {\bf 375/376} (1987), 104-156.

% \bibitem[Co]{Col} P. Colmez, {\em Fonctions L $p$-adiques}, S\'eminaire Bourbaki {\bf 851}, 1998/99.

\bibitem[Cox]{Cox}
David A. Cox, {\em Primes of the form $x^2+ny^2$: Fermat, Class Field Theory and Complex Multiplication}, John Wiley \& Sons, Inc. (1989)

%\bibitem[Dar1]{darmon-hpxh}
%H.~Darmon, {\em  Integration on $\cH_p\times \cH$ and arithmetic applications}. 
%Ann. of Math. (2) 154 (2001), no. 3, 589--639.

\bibitem[Dar2]{Dar-book}
H.~Darmon, {\em   Rational points on modular elliptic curves}, CBMS Regional Conference Series in Mathematics, 101. Published for the Conference Board of the Mathematical Sciences, Washington, DC; by the Americal Mathematical Society, Providence, RI, (2004)

\bibitem[DD]{DD}
H.~Darmon, S.~Dasgupta, {\em Elliptic units for real quadratic fields}, Annals of Mathematics (2) {\bf 163} (2006), no. 1, 301-346

%\bibitem[DDLR]{arizona}
%H.~Darmon, M.~Daub, S.~Lichtenstein, and V.~Rotger. 
%(With an appendix by W.~Stein) 
%{\em Algorithms for Chow-Heegner points via iterated integrals}, submitted.

\bibitem[DLR]{DLR}
H.~Darmon, A.~Lauder, and V.~Rotger. {\em
Stark points and $p$-adic iterated integrals attached to modular forms of weight one}, Forum Math.,  Pi (2015), Vol. 3, e8, 95 pages.



\bibitem[DP]{DP}
H. Darmon, R. Pollack, {\em
The efficient calculation of Stark-Heegner points via overconvergent modular symbols},  
Israel Journal of Mathematics, {\bf 153} (2006), 319-354. 



\bibitem[DR1]{DR1}
H.~Darmon and V.~Rotger, {\em 
Diagonal cycles and Euler systems I: A p-adic Gross-Zagier formula}. 
Annales Scientifiques de l'Ecole Normale Sup\'erieure, {\bf 47} no. 4 (2014), 779-832.

\bibitem[DR2]{DR2}
H.~Darmon and V.~Rotger, {\em 
Diagonal cycles and Euler systems II: $p$-adic families  and the Birch and Swinnerton-Dyer conjecture}, to appear in the Journal of the Amer. Math. Soc.


%\bibitem[DRS]{DRS}
%H.~Darmon, V.~Rotger, and I.~Sols. 
%{\em Iterated integrals, diagonal cycles, and rational points on elliptic curves}. 
%Publications Math\'ematiques de Besan\c{c}on, vol.2, 2012, pp. 19-46. 

%\bibitem[Das]{Das-thesis} S. Dasgupta, {\em Stark's conjectures}, Harvard Master thesis, 1999.

%\bibitem[DS]{serre1} P.~Deligne and J.-P.~Serre,
%{\em Formes modulaires de poids $1$}, Ann. Sci. Ecole Norm. Sup. (4) {\bf 7} (1974) (1975), 507--530. 


\bibitem[deS]{dS} 
E. de Shalit,  {\em Iwasawa theory of elliptic curves with complex multiplication. $p$-adic $L$-functions}, 
{\em Perspectives in Mathematics} {\bf 3}, Academic Press, Inc., Boston, MA (1987).


 
% \bibitem[DiSh]{DS} F. Diamond and J. Shurman, A First Course in Modular Forms, Springer GTM 228.


%\bibitem[Die]{Dieu} L. V. Dieulefait, {\em Langlands Base Change for $\GL(2)$}, Annals Math. {\bf 176} (2012), 1015--1038.     

\bibitem[Do]{Dok}
V. Dokchitser, {\em $L$-functions of non-abelian twists of elliptic curves}, Cambridge Ph.D dissertation, 2005.

%\bibitem[DL]{Duke-Li} 
%W. Duke and Y. Li, {\em Mock modular forms of weight one}. Preprint.

% \bibitem[FeGr]{FeGr} B. Ferrero, R. Greenberg, {\em On the behavior ...}, Invent. Math.

%\bibitem[GR]{GR} J. Gonz\'alez, V. Rotger, {\em Non elliptic Shimura curves of genus one}, J. Math. Soc. Japan {\bf 58:4} (2006), 927--948.

%\bibitem[Gre]{Gre} M. Greenberg, ...

\bibitem[Gro]{Gro}
B. Gross, {\em Heegner points on $X_0(N)$}, Modular forms (Durham, 1983), Ellis Horwood Ser. Math. Appli.: Statist. Oper. Res., Horwood, Chichester, (1984), pp. 87-105.

\bibitem[Gro2]{Gr}
B. Gross, {\em On the factorization of $p$-adic L-series}, Invent. Math. {\bf 57}, No. 1 (1980), 83--95.

\bibitem[GZ]{GZ}
B. Gross, D. Zagier, {\em Heegner points and derivatives of $L$-series}, Invent. Math. {\bf 84} (1986), 225--320.

%\bibitem[GM1]{GM12} X. Guitart, M. Masdeu, {\em Elementary matrix decomposition and the computation of Darmon points with higher conductor}

%\bibitem[GM2]{GM13} X. Guitart, M. Masdeu, {\em The effective computation of quaternionic p-adic Darmon points},

%\bibitem[HaKu]{HaKu} M. Harris, S. Kudla, {\em The central critical value of a triple product $L$-function},
%Ann. Math. (2) {\bf 133} (1991), 605--672.


%\bibitem[Ha]{Harron-CM} R. Harron, {\em The exceptional zero conjecture for symmetric powers of CM modular forms: the ordinary case}, Int. Math. Res. Notices, to appear.

\bibitem[Hi1]{Hi81} H. Hida, {\em Congruences of cusp forms and special values of their zeta functions}, Invent. Math. {\bf 63}, 225-261 (1981).

%\bibitem[Hi2]{Hi88}  H. Hida, {\em A $p$-adic measure attached to the zeta functions associated with two
%elliptic modular forms II}, Ann. Inst. Fourier (Grenoble) {\bf 38} (1988) 1--83.

%\bibitem[Hi3]{Hi90}  H. Hida, {\em p-adic $L$-functions for base change lifts of $\GL_2$ to $\GL_3$}, Perspectives in Mathematics {\bf 11}, II (1990), 93--142.

\bibitem[Hi4]{hida-book}
H. Hida, Elementary theory of $L$-functions and Eisenstein series,
 {\em London Math. Soc. St. Texts} {\bf 26} (1993).


%\bibitem[Ja]{Jac}
%H. Jacquet, Automorphic forms on GL(2), Part II, {\em Lecture Notes Math.} {\bf 278} Springer,
%Berlin-New York, 1972.

\bibitem[Kani1]{Kani1} 
E. Kani, {\em The space of binary theta series}, Ann. Sci. Math. Quebec {\bf 36} (2012), 501-534


\bibitem[Kani2]{Kani2}
E. Kani, {\em Binary theta series and modular forms with complex multiplication}, to appear in Int. J. Number Theory.

%\bibitem[Kato]{kato}


\bibitem[Katz1]{Katz1}
N. M. Katz, {\em $p$-adic interpolation of real analytic Eisenstein series}, Annals Math. {\bf 104}, no.\,3 (1976), 459-571.



\bibitem[Katz2]{Katz2}
N. M. Katz, {\em $p$-adic $L$-functions for CM fields}, Invent. Math. {\bf 49} (1978), 199--297.


\bibitem[KW]{KW} C. Khare, J.-P. Wintenberger, {\em Serre's modularity conjecture (I)}, Invent. Math. {\bf 178} (2009), 485--504. 

\bibitem[La1]{lauder-alg}
A.~Lauder. {\em Computations with classical and p-adic modular forms}, 
LMS J. Comput. Math. {\bf 14} (2011) 214-231.
 
\bibitem[La2]{lauder}
A.~Lauder. {\em Efficient computation of Rankin $p$-adic $L$-functions}, in Computations with Modular Forms, Proceedings of a Summer School and Conference, Heidelberg, August/September 2011, Boeckle G. and Wiese G. (eds), Springer Verlag (2014), 181-200.

\bibitem[MTT]{MTT} B. Mazur, J. Tate, J. Teitelbaum, On $p$-adic analogues of the conjectures of Birch and Swinnerton-Dyer, \emph{Invent. Math.} {\bf 84} (1986), no. 1, 1--48.


%\bibitem[Ne]{Nek}
%J.~Nekov\'a\v{r}, {\em  On the $p$-adic height of Heegner cycles}, Math. Ann. {\bf 302} (1995), 
%no. 4, 609--686. 


%\bibitem[PR1]{perrin-riou87}
%B. Perrin-Riou, {\em D\'eriv\'ees de fonctions L p-adiques et points de Heegner}, 
%Invent. Math. {\bf 89} (1987), 456-510. 


%\bibitem[PR2]{perrin-riou93}
%B.~Perrin-Riou, 
%{\em Fonctions $L$ $p$-adiques d'une courbe elliptique et points rationnels}. 
%Ann. Inst. Fourier (Grenoble)  {\bf 43}  (1993),  no. 4, 945--995. 



\bibitem[Pet49]{Pe} H. Petersson, {\em Uber die Berechnung der skalarprodukte ganzer modulformen}, Comm. Math. Helv. {\bf 22} (1949), 168-199 .

\bibitem[Pr]{Pr} D. Prasad, {\em Trilinear forms for representations of $\GL(2)$ and local epsilon factors}, Compositio Math. {\bf 75} (1990), 1--46.



\bibitem[Rob]{GR73} G. Roberts, Unites elliptiques et formules pour le nombre de classes des extensions abeliennes d'un corps quadratique
imaginaire, Bull. Soc. Math. France, M\'emoire {\bf 36}, (1973), 77 pp.


\bibitem[Roh]{Roh}
D. Rohrlich, {\em Root numbers}, notes from PCMI/IAS: Arithmetic of L-functions at Park City Mathematics Institute, Park City, Utah, June 29-July 17, (2009) 



\bibitem[Ru1]{rubin}
K.~Rubin, {\em  $p$-adic $L$-functions and rational points on elliptic curves with 
complex multiplication}. Invent. Math. {\bf 107} (1992), no. 2, 323--350. 


\bibitem[Ru2]{rubin-bu}
K.~Rubin, {\em $p$-adic variants of the Birch and Swinnerton-Dyer conjecture for 
elliptic curves with complex multiplication}, in  $p$-adic monodromy 
and the Birch and Swinnerton-Dyer conjecture (Boston, MA, 1991), 71--80,
Contemp. Math., {\bf 165}, Amer. Math. Soc. (1994), Providence, RI.


% \bibitem[Sc]{Sc}  C. -G. Schmidt, {\em $p$-adic measures attached to automorphic representations of $\GL(3)$}, Inventiones Math. {\bf 92} (1988), 597--631.


\bibitem[Sh76]{Sh76} 
G. Shimura, {\em The special values of the zeta functions associated with cusp forms}, Comm, Pure. Appl. Math. XXIX (1976), 783-804.


\bibitem[Sil94]{Sil94}
J. H. Silverman, {\em Advanced topics in the arithmetic of elliptic curves}, Graduate texts in mathematics (1994), 151.
 
%\bibitem[St]{stark1}
%H.M. Stark, {\em $L$-functions at $s = 1$. II. Artin $L$-functions with Rational Characters}, Advances Math.
%{\bf 17} (1975), 60--92.

%\bibitem[Ta]{Tate-Stark} J. Tate, Stark conjectures.

\end{thebibliography}


\title[Stark points and Hida-Rankin $p$-adic $L$-function]
{Stark points and Hida-Rankin $p$-adic $L$-function}



\author{Daniele Casazza and Victor Rotger}

\begin{abstract}
This article is devoted to the {\em elliptic Stark conjecture} formulated in \cite{DLR}, which proposes a formula for the transcendental part of a $p$-adic avatar of the leading term at $s=1$ of the Hasse-Weil-Artin $L$-series $L(E,\varrho_1\otimes \varrho_2,s)$ of an elliptic curve $E/\Q$ twisted by the tensor product $\varrho_1\otimes \varrho_2$ of two odd $2$-dimensional Artin representations, when the order of vanishing is two. The main ingredient of this formula is a $2\times 2$ $p$-adic regulator involving the $p$-adic formal group logarithm of suitable Stark points on $E$. This conjecture was proved in \cite{DLR} in the setting where $\varrho_1$ and $\varrho_2$ are induced from characters of the same imaginary quadratic field $K$. In this note we prove a refinement of this result, that was discovered experimentally in \cite[Remark 3.4]{DLR} in a few examples. Namely, we are able to determine the algebraic constant up to which the main theorem of \cite{DLR} holds in a particular setting where the Hida-Rankin $p$-adic $L$-function associated to a pair of Hida families can be exploited to provide an alternative proof of the same result. This constant encodes local and global invariants of both $E$ and $K$.
\end{abstract}

\address{D. C.: Universit\'{e} de Bordeaux, Bordeaux, France; Universitat Polit\'{e}cnica de Catalunya, Barcelona, Spain}
\email{daniele.casazza@u-bordeaux.fr}
\address{V. R.: Departament de Matem\`{a}tica Aplicada II, Universitat Polit\`{e}cnica de Catalunya, C. Jordi Girona 1-3, 08034 Barcelona, Spain}
\email{victor.rotger@upc.edu}

%\subjclass{11G18, 14G35}


\maketitle

\tableofcontents







\section{Introduction}

Let $E/\Q$ be an elliptic curve of conductor $N_E$ and let $f\in S_2(N_E)$ denote the eigenform associated to it by modularity. Let in addition
\begin{equation*}
	\varrho : G_\Q \lra \GL(V_\varrho) = \GL_n(L)
\end{equation*}
be an Artin representation with values in a finite extension $L/\Q$ and factoring through the Galois group $\Gal(H/\Q)$ of a finite Galois extension $H/\Q$. Here $V_\varrho$ is the $L[G_\Q]$-module underlying the Galois representation $\varrho$. 

We define the $\varrho$-isotypical component of the Mordell-Weil group of $E(H)$ as
$$
E(H)^\varrho = \Hom_{\Gal(H/\Q)}(V_\varrho, E(H) \otimes L),
$$
and set 
$$
 r(E,\varrho) := \dim_L E(H)^\varrho.
$$

Let $L(E,\varrho,s)$ denote the Hasse-Weil-Artin $L$-series  of the twist of $E$ by $\varrho$. This $L$-function is expected to admit analytic continuation to the whole complex plane and to satisfy a functional equation relating the values at $s$ and $2-s$, although this is only known in a few cases, including the ones we consider in this note. Assuming these properties, we may define the {\em analytic rank} of the pair $(E,\varrho)$ as
$$
r_{\an}(E, \varrho) := \ord_{s=1} L(E,\varrho,s).
$$

The equivariant refinement of the Birch and Swinnerton-Dyer conjecture (cf.\,\cite{Dok},\cite{Roh}) for the twist of $E$ by $\varrho$ predicts that
\begin{equation} \label{equivariantBSD}
r(E,\varrho) \stackrel{?}{=} r_{\an}(E, \varrho).
\end{equation}

The equality \eqref{equivariantBSD} is known to be true in rather few cases, always under the assumption that $r_{\an}(E,\varrho) = 0$ or $r_{\an}(E,\varrho) = 1$. We refer to \cite{BDR2} and \cite{DR2} for the latest developments in this direction. In particular one is often at a loss to construct non-trivial points in $E(H)^\varrho$.

In the recent work  \cite{DLR}, Darmon, Lauder and Rotger propose a new approach for computing (linear combinations of logarithms of) non-zero elements in $E(H)^\varrho$ for a wide class of Artin representations $\varrho$ under the assumption that $r_{\an}(E, \varrho)\leq 2$. More precisely, $\varrho$ is allowed to be any irreducible constituent of the tensor product 
$$
\varrho = \varrho_1 \otimes \varrho_2
$$
of any pair of odd, two-dimensional Galois representations $\varrho_1$ and $\varrho_2$ of conductors $N_1$ and $N_2$ respectively, satisfying $$(N_E,N_1N_2)=1, \quad \det(\varrho_2) = \det(\varrho_1)^{-1} \quad \mbox{and} \quad r_{\an}(E, \varrho_1\otimes \varrho_2)=2.$$ After the ground-breaking works of Buzzard-Taylor, Khare-Wintenberger and others, we now know that these representations are modular and hence $\varrho_1$ and $\varrho_2$ are isomorphic to the Deligne-Serre Artin representation associated to eigenforms
$$
g\in M_1(N,\chi)\, \quad h\in M_1(N,\chi^{-1}),
$$
respectively, where the level $N$ might be taken to be the least common multiple of $N_E$, $N_1$ and $N_2$, and the nebentype $\chi = \det(\varrho_1)$ is the determinant of $\varrho_1$ regarded as a Dirichlet character.

Fix a prime $p\nmid N$ at which $g$ satisfies the classicality hypotheses C-C' introduced in \cite{DLR}.
The main conjecture of \cite{DLR} is a formula relating
\begin{itemize}

\item a $p$-adic iterated integral associated to the triple $(f,g,h)$ of eigenforms, to

\item an explicit linear combination of formal group logarithms of points of infinite order on $E(H)$, and

\item the $p$-adic logarithm of a Gross-Stark unit associated to the adjoint of $g$.


\end{itemize}

As shown in \cite[\S 2]{DLR}, the $p$-adic iterated integral can be recast as the value of the triple-product Harris-Tilouine $p$-adic $L$-function constructed in \cite[\S 4.2]{DR1} at a point of weights $(2,1,1)$. In this paper we place ourselves in a setting where $h$ is Eisenstein and Harris-Tilouine's $p$-adic $L$-function alluded to above can be replaced with the more standard Rankin $p$-adic $L$-function of Hida \cite[\S 7.4]{hida-book}. This observation is crucial for our purposes, for the latter $p$-adic $L$-function is more amenable to explicit computations.

The linear combination of logarithms of points mentioned above arises as the determinant of a $2 \times 2$ matrix introduced in \cite{DLR} that plays the role of a $p$-adic avatar of the regulator in the classical setting. Note however that this $p$-adic regulator does not coincide with the one considered by Mazur, Tate and Teitelbaum in \cite{MTT}, as the $p$-adic height function is replaced in \cite{DLR} with the formal logarithm on the elliptic curve. 

In some instances, as we shall see below, the logarithm of one of the points may be isolated in such a way that the conjectural expression suggested in \cite[Conjecture ES]{DLR} gives rise to a formula that allows to compute the point in terms of more accessible quantities which include the $p$-adic iterated integral and logarithms of global units and points that are rational over smaller number fields.

To describe more precisely our main result, 
%assume $N_E\geq 7$ is square-free and $N_E\equiv 3\,(\mathrm{ mod }\,4)$.  
let $K$ be an imaginary quadratic field of discriminant $-D_K$ with $D_K \geq 7$ and let $\cO_K$ denote its ring of integers. Let also $h_K = |\Cl_K|$ denote the class number, $g_K = [ \Cl_K : \Cl_K^2 ]$ be the number of genera, and $\chi_K$ be the quadratic Dirichlet character associated to $K/\Q$. 

Let $c\geq 1$ be a fixed positive integer relatively prime to $N_E$ and let $H/K$ denote the ring class field of $K$ of conductor $c$. Let 
$$\psi: \Gal(H/K) \,\lra \, \bar\Q^\times$$ be a character of finite order and
\begin{equation*}
g:=\theta(\psi) \in M_1(D_K c^2,\chi_K)
\end{equation*}
be the  theta series associated to $\psi$. This is an eigenform of level $D_Kc^2$ and nebentype character $\chi_K$. The form $g$ is cuspidal if and only if $\psi^2\ne 1$.


Let $\Q_{\psi}$ denote the finite extension generated by the values of $\psi$. 
Let $\varrho_\psi = \mathrm{Ind}_\Q^K(\psi)$ denote the odd two-dimensional Artin representation of $G_\Q$ induced by $\psi$ and write $V_{\psi}$ for the underlying two-dimensional $\Q_{\psi}$-vector space.


Fix now an odd rational prime $p=\wp \bar\wp\nmid N_E D_K c^2$ that splits in $K$. 
The choice of the ideal $\wp$ above $p$ determines a Frobenius element $\Fr_{p}$ in $\Gal(H/K)$. The eigenvalues of $\Fr_p$ acting on $V_{\psi} $ are
$$
\alpha = \psi(\Fr_p), \quad \beta= \bar\psi(\Fr_p) = \alpha^{-1}.
$$

As mentioned above, we assume that $g$ satisfies the classicality hypotheses C-C' of \cite{DLR} at the prime $p$. When $g$ is cuspidal, i.e.\,$\psi^2\ne 1$, one can verify\footnote{Indeed, when $g$ is cuspidal this hypothesis not only asks  that $\alpha \ne \beta$ but also that there should exist no real quadratic field $F$ in which $p$ splits such that $\varrho_g \simeq \Ind_F^\Q(\xi)$ for some character $\xi$ of $F$. However, in our CM setting, the existence of a character $\xi$ of a real quadratic field $F$ such that $\mathrm{Ind}_\Q^K(\psi)\simeq \mathrm{Ind}_\Q^F(\xi)$ implies that $\Gal(H/K)\simeq C_4$. Then $F$ is the single real quadratic field contained in the quadratic extension of $K$ cut out by $\psi^2$, and the condition $\alpha\ne \beta$ implies that $p$ can not split in $F$.} that in our setting the hypothesis is equivalent to assuming that $\alpha \ne \beta$. When $\theta(\psi)$ is Eisenstein, i.e.\,$\psi^2 = 1$, one expects\footnote{Indeed, when $\theta(\psi)$ is Eisenstein, it is shown in \cite[\S 1]{DLR} that a necessary condition for Hyptheses C-C' to hold is that $\alpha = \beta$, but this is automatically satisfied because $\psi^2=1$. As explained in loc.\,cit.,\,this is also expected to be a sufficient condition.} that hypotheses C-C' are always satisfied in our case.



\begin{remark}\label{assBD}
This assumption might appear unmotivated when one encounters it for the first time. In the case where $g$ is cuspidal, a striking recent result of Bella\"iche and Dimitrov \cite{BeDi} shows that hypotheses C-C' ensure that the points in the eigencurve associated to either of the two ordinary $p$-stabilisations $g_\alpha$, $g_\beta$ of $g$ is smooth and the weight map is \'etale at these points. 

Building on their work, it was further shown in \cite[\S 1]{DLR} that this assumtion implies that all overconvergent generalised eigenforms with the same system of Hecke eigenvalues as $g_\alpha$ are necessarily classical. This in turn guarantees that the $p$-adic iterated integral we will consider in \eqref{iterated} below is well-defined. 

Finally, it is also worth mentioning that hypotheses C-C' imply that the elliptic unit $u_{\psi^2}$ introduced in \eqref{ell-unit} below may be characterized, up to multiplication by scalars in $\Q_{\psi}^\times$, as follows: Up to $\Q_{\psi}^\times$ there is a single $\Gal(H/K)$-equivariant homomorphism $\varphi: \Q_{\psi}(\psi^2) \lra \cO_H^\times \otimes \Q_{\psi}$ and $u_{\psi^2}$ spans $\mathrm{Im}(\varphi)$.
\end{remark}


The choice of $\wp$ determines an embedding $K\hookrightarrow \Q_p$, that we extend to an embedding $H\hookrightarrow \C_p$. Write $H_p$ for the completion of $H$ with respect to this embedding.

Note that there is a natural isomorphism of $\Gal(H/K)$-modules $$E(H)^{\varrho_\psi} = E(H)^\psi \oplus E(H)^{\bar\psi}$$ where
$$
E(H)^{\psi} = \{ P\in E(H)\otimes \Q_{\psi}: \, P^\sigma = \psi(\sigma)P \, \mbox{ for all }\,\sigma\in \Gal(H/K)\},
$$
and $E(H)^{\bar\psi}$ is defined likewise. 

%Let us define:
%\begin{equation*}
%	\psi_1 := \psi_g\psi_h, \qquad \psi_2 := \psi_g \psi_h'
%\end{equation*}
%where $\psi'$ denotes the character obtained by applying to $\psi$ the non trivial automorphism in $\Gal(K|\Q)$, i.e. $\psi'(\sigma)= \psi ( \sigma_0\sigma \sigma_0^{-1})$, where $\sigma_0 \in G_\Q \setminus G_K$. They are both ring class characters of $K$ associated %with the orders $\cO_{c_1}$ and $\cO_{c_2}$. For $\psi_g$ (and similarly for $\psi_h$), if we write $V_{\psi_g} := \Ind_\Q^K \psi_g$ for the vector space underlying $\varrho_g$, using Artin's formalism one can see that we have the following decompositions:
%\begin{equation*}
%	V_{gh} := V_{\psi_g}\otimes V_{\psi_h}= V_{\psi_1}\oplus V_{\psi_2},  \qquad L(E, \varrho_{gh}, s) = L(E, \psi_1,s)L(E,\psi_2,s)
%\end{equation*}
%The field of coefficients is then generated by the traces of $V_{\psi_1}$ and $V_{\psi_2}$ and it is contained in $\Q(\psi_g, \psi_h)$.

\vspace{0.2cm}

Throughout we impose the classical {\em Heegner hypothesis}:

\begin{ass}
There exists an integral ideal $\mathfrak N$ in $\cO_K$ such that $\cO_K/\mathfrak N \simeq \Z/N_E\Z$. 
\end{ass}

This assumption guarantees the existence of non-trivial {\em elliptic units} in $H^\times$ and {\em Heegner points} in $E(H)$ arising from the classical modular curve $X_0(N_E)$. In particular, there exists a canonical unit  (cf.\,e.g.\,\cite[\S 1]{DD},\cite[p.\,15-16]{GR73})
associated to the character $\psi^2$ defined as
\begin{equation}\label{ell-unit}
u_{\psi^2}=\begin{cases}
\mbox{any } p\mbox{-unit in } \cO_H[\frac{1}{p}]^\times \,\mbox{ satisfying } \, (u_\wp) = \wp ^{h_K}, & \mbox{ if } \,\psi^2 = 1, \\
\sum_{\sigma \in \Gal(H/K)} \psi^2(\sigma) u^\sigma \in \cO_H^\times\otimes \Q_{\psi}, & \mbox{ if } \,\psi^2\ne 1.
\end{cases}
\end{equation}

Likewise, 
associated to $\psi$ there is the canonical point (cf.\,e.g.\,\cite[{\bf I.} \S 6]{GZ})
\begin{equation}\label{Heegner-point}
%P_K := \sum_{\sigma \in \Gal(H/K)} x^\sigma \in E(K), \quad 
P_\psi := \sum_{\sigma \in \Gal(H/K)} \psi(\sigma) x^\sigma \in E(H)^{\psi^{-1}} \subset E(H)^{V_\psi}.
\end{equation}

In \eqref{ell-unit} and \eqref{Heegner-point}, $u$ and $x$ are arbitrary choices of an elliptic unit and Heegner point with CM by $\cO_K$, respectively. Since, up to sign, all such choices lie in the same orbit under $\Gal(H/K)$, $u_{\psi^2}$ and $P_\psi$ do not depend on this, again up to sign. Notice that the convention is different to the one usually taken in literature in the sense that $\sigma$ acts on $P_\psi$ as $\psi(\sigma)^{-1}$ instead of $\psi(\sigma)$.

Set $N= \mathrm{lcm}(D_K c^2,N_E)$. Write $S_2(N)[f]$ (resp.\,$M_1(N,\chi_K)[g]$) for the subspace of $S_2(N)$ (resp.\,of $M_1(N,\chi_K)$) consisting of modular forms that are eigenvectors for all good Hecke operators $T_\ell$, $\ell\nmid N$, with the same eigenvalues as $f$ (resp. $g$). 

Fix a modular form $\testf = \sum_{n\geq 1} a_n(\testf)q^n\in S_2(N)[f]$ and let
$$
\testf^{[p]} = \sum_{p\nmid n} a_n(\testf)q^n
$$
denote the $p$-depletion of $\testf$.

Fix as well a modular form $\testg \in M_1(N,\chi_K)[g]$ and let
$$
\testg_\alpha(q) = \testg(q)-\beta_g \testg(q^p)\in M_1(pN,\chi_K)
$$
denote the ordinary stabilisation of $\testg$ on which $U_p$ acts with eigenvalue $\alpha$.

Associated to the pair $(\testf,\testg_\alpha)$ we may define a $p$-adic iterated integral as follows. Let $
h:=\mathrm{Eis}_1(1,\chi_{K})\in M_1(D_K,\chi_K)$ be the Eisenstein series  associated to the Dirichlet character $\chi_K$, as defined e.g.$\,$in \cite[\S 2.1.2]{BDR1}. Let also
\begin{equation}\label{h}
	\testh:= E_{1,\chi_{_{K,N}}}\in M_1(N,\chi_K)[h]
\end{equation}
denote the Eisenstein series of level $N$ in the isotypic eigenspace of $h$ that we introduce in \eqref{trueEisenstein} below. Let 
$$
e_{\text{ord}}: M^{\oc}_1(N,\chi_K) \lra M^{\oc, \ord}_1(N,\chi_K)
$$ 
denote Hida's ordinary idempotent on the space of overconvergent modular forms of weight $1$, tame level $N$ and tame character $\chi_K$. Let $$
e_{g_\alpha}^*: M^{\oc, \ord}_1(N,\chi_K) \lra M^{\oc,\ord}_1(N,\chi_K)[[g^*_\alpha]]
$$ 
denote the projection onto the generalised eigenspace attached to $g_\alpha^*$. 

Letting  $d=q\frac{d}{dq}$ denote Serre's $p$-adic derivative operator, the overconvergent ordinary modular form
$$
e_{g_\alpha}^* e_{\text{ord}}(d^{-1}\testf^{[p]}\times E_{1,\chi_{_{K,N}}})
$$
lies in the space of classical modular forms $M_1(pN,\chi_K)[g_\alpha^*]$ of weight $1$, level $pN$ and nebentype $\chi_K$, with coefficients in $\Q_p$, as explained in Remark \ref{assBD}.


Let $\testgamma_{g_\alpha}$ be an element in the $\Q_{\psi}$-dual space of $M_1(pN,\chi_K)[g^*_\alpha]$; more specifically, we take it to be the one associated to $\testg_\alpha$ in \cite[Proposition 2.6]{DLR}. By extending scalars, we may regard $\testgamma_{g_\alpha}$ as a $\Q_p(\psi)$-linear functional on $M_1(pN,\chi_K)[g^*_\alpha]$ as well.

Following \cite{DLR}, define
\begin{equation}\label{iterated}
\int_{\testgamma_{g_\alpha}} \testf \cdot E_{1,\chi_{_{K,N}}} := \testgamma_{g_\alpha} (e_{g_\alpha}^* e_{\text{ord}}(d^{-1}\testf^{[p]}\times E_{1,\chi_{_{K,N}}})) \in \C_p.
\end{equation}

In \cite[Theorem 3.3]{DLR} it was proved a statement that, specialised to our setting, asserts the following. Let $\log_p$ denotes the usual branch of the $p$-adic logarithm on $H_p^\times$ that satisfies $\log_p(p)=0$, and let $\log_{E,p}$ denote the formal group logarithm on $E/H_p$. 

\begin{theorem} \label{mainDLR}
Assume $r_{\an}(E/K,\psi)=1$. There exists a finite extension $L$ of $\Q_{\psi}$ and a scalar $\lambda(\testf,\testg) \in L$ such that
$$
\int_{\testgamma_{g_\alpha}} \testf \cdot E_{1,\chi_{_{K,N}}} = \lambda(\testf,\testg)\cdot \frac{\log^2_{E,p}(P_\psi)}{\log_p(u_{\psi^2})}.
$$
Moreover, there is a suitable choice of $\testf$ and $\testg$ such that $\lambda(\testf,\testg)\ne 0$.
\end{theorem}


Several questions arise naturally in the light of the above statement: 
\begin{itemize}
\item Can the field $L$ be determined?

\item Are there explicit choices of $\testf$ and $\testg$ for which $\lambda(\testf,\testg)\ne 0$?

\item Can the scalar $\lambda(\testf,\testg)$ be computed explicitly? 


\item Does $\lambda(\testf,\testg)$ have any arithmetical meaning?
\end{itemize}


The aim of this note is answering these questions by proving an explicit formula for the scalar $\lambda(\testf,\testg)$ in terms of local and global arithmetic invariants of $E$ and $\psi$. In doing this, we prove as a particular case a formula that was already conjectured and verified numerically in \cite[Remark 3.4 and (45)]{DLR}.

While the main conjecture of \cite{DLR} may be regarded as a $p$-adic analogue of the {\em rank part} of the classical equivariant Birch and Swinnerton-Dyer conjecture, our main result provides a precise formula for the {\em leading term} in the particular setting we have placed ourselves. Hence Theorem \ref{main-theta} below may be regarded as a $p$-adic avatar of the formula for the leading term predicted by the conjecture of Birch and Swinnerton-Dyer, and  we hope it may suggest a $p$-adic variant of the classical {\em equivariant Tamagawa number conjecture}; more details on this may appear elsewhere.

The reader may also find Theorem \ref{main-theta} interesting from the computational point of view, as it provides an explicit $p$-adic formula for the Heegner point $P_\psi$ in $E(H_p)/E(H_p)_{\mathrm{tors}}$. Namely,
\begin{equation}
P_\psi = \exp_{E,p} \left( \sqrt{\frac{\log_p(u_{\psi^2})}{\lambda(\testf, \testg)}\cdot \int_{\testgamma_{g_\alpha}} \testf \cdot E_{1,\chi_{_{K,N}}}} \, \right).
\end{equation}

Let $\Q(f_N)$ denote the finite extension of $\Q$ generated by the roots of the Hecke polynomials $T^2-a_q(f)T+q$ for all primes $q\mid N$, $q\nmid N_E$. Note that if $\testf \in S_2(N)[f]$ is chosen to be a normalized eigenvector for all good {\em and bad} Hecke operators $T_\ell$ for all primes $\ell$, then the fourier coefficients of $\testf$ lie in $\Q(f_N)$. In a similar way, observe also that  if $\testg \in M_1(N)[g]$ is chosen to be an eigenvector for all good and bad Hecke operators, then the fourier coefficients of $\testg$ lie in $\Q_\psi$. Write $\Q_{\psi}(f_N)$ for the compositum of $\Q(f_N)$ and $\Q_{\psi}$.

\begin{theorem}\label{main-theta} 
\begin{enumerate}
\item[(i)] If $\testf$ and $\testg$ are chosen to be eigenvectors for all good and bad Hecke operators, then $L$ can be taken to be $\Q_\psi(f_N)$ and $\lambda(\testf,\testg)\ne 0$.


\item[(ii)] Assume that $D_K=N_E$ and $c=1$. Then the following formula holds true for $\testf=f$, $\testg=g$:
$$
\lambda(f,g) =  \frac{(p- a_p(f)\psi(\bar\wp)  + \psi^{2}(\bar\wp) )^2}{p} \cdot \frac{\lambda_0}{h_K g_K}  
$$
where
$$
\lambda_0 = \begin{cases} \frac{1}{p-1}   & \quad \mbox{if } \psi^2=1, \mbox{ that is to say, if } g \mbox{ is Eisenstein} \\
\frac{12}{p-(p+1)\psi^{-2}(\bar \wp)+\psi^{-4}(\bar \wp) } & \quad \mbox{if } \psi^2\ne 1, \mbox{ that is to say, if } g \mbox{ is cuspidal.}
\end{cases}	
$$
\end{enumerate}
\end{theorem}

Note that in the special case in which $N_E$ is prime and $\psi = 1$, we obtain
\begin{equation} \label{christmas}
\lambda(f,g) = \frac{|E(\F_p)|^2}{p(p-1)h_K}.
\end{equation}
As reported in \cite[Remark 3.4 and (45)]{DLR}, this formula was verified numerically in several examples, and here it is proved unconditionally.



%Using the Rankin-Selberg method, one can relate a special value of the Hida $p$-adic Rankin $L$-function to the $p$-adic iterated integrals and then, in a way not far from the article \cite{DLR}, we describe a functional equation which relates those $p$-adic $L$-functions to other $p$-adic $L$-functions, in particular Bertolini-Darmon-Prassana $p$-adic $L$-function, which provides a link with the elliptic logarithm of Heegner points. The main result we are going to prove is the following:


The proof of Theorem  \ref{main-theta} actually provides an alternative proof of Theorem \ref{mainDLR} in the setting considered here. As in loc.\,cit.\,we compare the values of several $p$-adic $L$-functions at several points lying outside the region of interpolation, the main novelty with respect to \cite{DLR} being that we exploit Hida-Rankin $p$-adic $L$-function associated to the convolution of two Hida families, instead of the triple-product Harris-Tilouine $p$-adic $L$-function associated to a triple of Hida families. Since the former has been extensively studied in the literature, this alternative approach allows us to perform the explicit computations that are needed in order to derive the sought-after refined formula.




\section{Hecke characters, theta series and Katz's $p$-adic $L$-function}\label{Hecke}

Let $K/\Q$ be an imaginary quadratic field of discriminant $-D_K$ and let $\fc \subset \cO_K$ be an integral ideal. Let $I_\fc$ denote the group of fractional ideals of $K$ that are coprime to $\fc$.

A \emph{Hecke character} of infinity type $(\ka_1,\ka_2)$ of $K$ is a homomorphism 
$$
\psi: I_\fc \lra \C^\times$$ 
such that
\[
	\psi( (\alpha) ) = \alpha^{\ka_1} \overline{\alpha}^{\ka_2}
\]
for all $\alpha \equiv 1 \pmod{\fc}$. The \emph{conductor} of $\psi$ is the largest ideal $\fc_\psi$ for which this holds. Let us introduce some basic notations and terminology:



\begin{itemize}
	\item The norm map $\mathbf N_K:= |N^K_{\Q}|: I_\fc \lra \C^\times$ gives rise to a Hecke character of infinity type $(1,1)$ and conductor $1$.

	\item For any Hecke character $\psi$ of infinity type $(\ka_1,\ka_2)$ define $\psi'(\mathfrak a) = \psi(\overline{\mathfrak a})$, where $\overline{x}$ denotes complex conjugation; we say that $\psi$ is \emph{self-dual}, or \emph{anticyclotomic}, if $\psi  \psi'= \mathbf N_K^{\ka_1+\ka_2}$.

	\item A Hecke character of finite order (or infinity type $(0,0)$) can be regarded as a character of $G_K = \Gal(\bar K/K)$ via class field theory. We continue to denote by $\psi$ the resulting character, which is anticyclotomic.

\item The \emph{central character} $\varepsilon_\psi$ of $\psi$ is the single Dirichlet character satisfying
$$\psi|_{\Q} = \varepsilon_\psi \mathbf N^{\ka_1+\ka_2}_K.$$

\end{itemize}

The following lemma is well-known.

\begin{lemma} \label{thm-Cox} Let $\psi$ be a Hecke character of finite order. The following are equivalent:
\begin{enumerate}

\item $\psi$ is a ring class character.

\item $\Ind_\Q^K(\psi)$ is a self-dual representation.

\item The central character of $\psi$ is trivial.
\end{enumerate}
\end{lemma}
%\begin{proof}
%The first two statements are equivalent by \cite[Theorem 9.18]{Cox}. That (2) implies (3) is obvious. In order to show that (3) implies (2), notice that $\cond (\varepsilon_\psi) \mid c$ and:
%	\[
%		\varepsilon_\psi(m) = \psi( (m) ) = 1, \qquad \forall m\equiv \pm 1 \pmod{c}.
%	\]
%	Moreover, for every prime $p \cO_K = \wp  \overline {\wp}$ split or ramified in $K$, coprime to $c$, we have:
% 	\[
%		\varepsilon(p) = \psi(\wp)\psi(\overline{\wp}) = 1.
%	\]
%	By the theorem on primes in arithmetic progression (e.g. \cite[Theorem 3.4]{Sil94}), for each $m$ coprime to $c$ we can find a degree one prime $p$ such that $m\equiv p \pmod{c}$. Then $\varepsilon_\psi(m) = \varepsilon_\psi(p) = 1$, so that $\varepsilon_\psi = 1_c$. {\bf To be kept in Daniele's thesis but not in this paper.}
%\end{proof}

Given a Hecke character of $K$ of infinity type $(\ka_1,\ka_2)$, we can associate to it a theta series as follows. Define the quantities $$a_n(\psi) = \sum_{\mathfrak a \in I_{\fc_\psi}^n} \psi(\mathfrak a),$$
where $I_{\fc_\psi}^n$ is the set of the ideals in $I_{\fc_\psi}$ whose norm is $n$. Define also $a_0(1) = h_K/w_K$ and $a_0(\psi) = 0$ otherwise. As shown in \cite{Kani1},
\begin{equation} \label{theta-definition}
	\theta_\psi := \sum_{n\geq 0} a_n(\psi) q^{n} = \sum_{n\geq 0} a_n(\theta_\psi) q^n \in M_{\ka_1+\ka_2}(D_K N^K_\Q(\fc_\psi), \chi_K \varepsilon_\psi)
\end{equation}
is the $q$-expansion of a normalized newform of weight $\ka_1+ \ka_2$, level $D_K N^K_\Q(\fc_\psi)$ and nebentype $\chi_K \varepsilon_\psi$. Moreover $\theta_\psi$ is Eisenstein if and only if $\psi = \psi'$; otherwise $\theta_\psi$ is a cusp form. 

Associated to $\psi$ and $\theta_\psi$ there are the Hecke $L$-functions 
\begin{align*}
L(\psi,s ) := \prod_{\mathfrak p} \left( 1-\frac{\psi(\mathfrak p )}{N^K_\Q \mathfrak p ^s} \right)^ {-1} \qquad \text{and} \qquad %= \sum_{\mathfrak a \subset \cO_k} \frac{\psi(\mathfrak a)}{N^K_\Q(\mathfrak a)^s}
L(\theta_\psi, s):=  \sum_{n\geq 1} \frac{a_n(\theta_\psi)}{n^s}.
\end{align*}

%\begin{equation*}
%L(\psi,s ) := \prod_{\mathfrak p} \left( 1-\frac{\psi(\mathfrak p )}{N^K_\Q \mathfrak p ^s} \right)^ {-1} = \sum_{\mathfrak a \subset \cO_k} \frac{\psi(\mathfrak a)}{N^K_\Q(\mathfrak a)^s} =  \sum_{n\geq 1} \frac{a_n(\theta_\psi)}{n^s} =: L(\theta_\psi, s)
%\end{equation*}
They can be extended to a meromorphic functions on $\C$. Since they coincide in a common region of convergence, they are actually the same function. From the definitions it is easy to verify that for any $k\in \Z$ we have
\begin{equation} \label{translation}
	L(\psi, s) = L(\psi \mathbf N_K^{k}, s+k).
\end{equation}


\subsection{Katz's two-variable $p$-adic $L$-function}\label{Katzs}

Assume $D_K\geq 7$ and let $\fc \subseteq \cO_K$ be an integral ideal. Fix a prime $p=\wp \bar\wp$ that splits in $K$. 

Denote by $\Sigma$ the set of Hecke characters of $K$ of conductor dividing $\fc$ and define $$\Sigma_K = \Sigma_K^{(1)}\cup \Sigma_K^{(2)}\subset \Sigma$$ to be the disjoint union of the sets
$$
\Sigma_K^{(1)}=\{ \psi \in \Sigma \mbox{ of infinity type } (\ka_1,\ka_2), \ka_1\leq 0, \ka_2\geq 1\},
$$
$$
\Sigma_K^{(2)}=\{ \psi \in \Sigma \mbox{ of infinity type } (\ka_1,\ka_2), \ka_1\geq 1, \ka_2\leq 0\}.
$$

For all $\psi \in \Sigma_K$, $s=0$ is a critical point for the Hecke $L$-function $L(\psi^{-1},s)$, and Katz's $p$-adic $L$-function is constructed by interpolating the (suitably normalized) values $L(\psi^{-1},0)$ as $\psi$ ranges over $\Sigma_K^{(2)}$. 

More precisely, let $\hat{\Sigma}_K$ denote the completion of $\Sigma_K^{(2)}$ with respect to the compact open topology on the space of functions on a certain subset of $\mathbf{A}^\times_K$, as described in \cite[\S 5.2]{BDP1}. By the work of Katz \cite{Katz1}, there exists a $p$-adic analytic function
$$
L_p(K): \hat{\Sigma}_K\lra \C_p
$$
which is uniquely characterized by the following interpolation property: for all $\psi \in \Sigma_K^{(2)}$ of infinity
%\footnote{In \cite[Thm.\,4.14]{dS}, he in fact asks the weights to satisfy $k_1>|k_2|$. (In his notation, $k_1=k$ and $k_2=j$.) This restriction is %not imposed in BDP for some reason...} 
type $(\ka_1,\ka_2)$,
\begin{equation}\label{int-Katz}
L_p(K)(\psi) = \mathfrak{e}_{K}(\psi) \mathfrak{f}_{K}(\psi) \frac{\Omega_p^{\ka_1-\ka_2}}{\Omega^{\ka_1-\ka_2}}L_\fc (\psi^{-1},0)
\end{equation}
where
\begin{itemize}
\item $L_\fc (\psi^{-1},s)$ is Hecke's $L$-function associated to $\psi^{-1}$ with the Euler factors at primes dividing $\fc$ removed,

\item $\Omega_p\in \C_p^\times$ is a $p$-adic period attached to $K$, as defined in \cite[(140)]{BDP1}, \cite[(25)]{BDP2},

\item $\Omega\in \C^\times$ is the complex period associated to $K$ as defined in \cite[(137)]{BDP1},

\item $\mathfrak{e}_{K}(\psi) = (1-\frac{\psi(\wp)}{p}) (1-\psi^{-1}(\bar\wp))$, and  $\mathfrak{f}_{K}(\psi) = \frac{(\ka_1-1)!\cdot D_K^{\ka_2/2}}{(2 \pi)^{\ka_2} }$.

\end{itemize}

%, which is equal to $2$ in our case since $D_K \ne 1, 3$.



%In particular, if the character $\psi$ is anticyclotomic (as is it in our case), then $L_p(K)(\psi) = L_p(K)(\psi\mathbf{N}_{K})$. 

%Indeed, recall that Gross' $p$-adic $L$-function differs from ours: up to a constant, $L_p^{Gr}(\psi) = L_p(K)(\psi^{-1})$. This implies that while $L_p^{Gr}$ satisfies a functional equation relating $\psi$ to $\hat\psi = (\psi')^{-1}\mathbf{N}_K^{-1} = \psi \mathbf{N}_K^{-1}$, our $p$-adic $L$-function relates then $\psi$ to $\psi \mathbf{N}_K$.

%The values of a $p$-adic $L$-function at points lying outside the region of interpolation are of great interest to us, as they may encode arithmetic invariants of the motives under consideration. One instance of this philosophy is %

The following result is commonly known as Katz's Kronecker $p$-adic limit formula. It computes the value of $L_p(K)$ at a finite order character $\psi$ of $G_K$, which lies outside the region of interpolation (cf.\,\cite[\S 10.4, 10.5]{Katz1}, \cite[p.\,90]{Gr}, \cite[Ch.\,II, \S 5.2]{dS}): 
\begin{equation}\label{KatzKronecker}
L_p(K)(\psi) = \mathfrak f_p(\psi) \cdot \log_p(u_{\psi^{-1}}),
\end{equation}
where
\begin{equation}\label{Katz-GZ}
\mathfrak f_p(\psi) % =  L_p(K)(\bar{\psi}_g\mathbf{N}_{K}) 
= \begin{cases}\frac{1}{2} (\frac{1}{p}-1)  & \mbox{ if } \psi=1 \\
 \frac{-1}{24c}  (1-\psi(\bar\wp)) (1-\frac{\psi(\bar\wp)}{p}) &  \mbox{ if } \psi\ne 1.
\end{cases}
\end{equation}
Here $c>0$ is the smallest positive integer in the conductor ideal of $\psi$.



\section{Classical and $p$-adic Rankin $L$-fuctions}

%In this section we essentially follow \cite{BD1} and \cite{BDR1} to describe Rankin's method in a %slightly more general setting.

\subsection{Eisenstein series}
Let $\chi : (\Z/N_\chi\Z)^\times \to \C$ be a Dirichlet character of conductor $N_\chi$ and let $\Q_\chi$ denote the finite extension of $\Q$ generated by the values of $\chi$. For any multiple $N$ of $N_\chi$ let $\chi_N$  denote the character mod $N$ induced by $\chi$. 

For every positive integer $k\geq 1$, let $M_k(N,\chi_N)$ and $S_k(N,\chi_N)$ denote the spaces of holomorphic (resp.\,cuspidal) modular forms of weight $k$, level $N$ and character $\chi_N$. 

We also let $M^{\an}_k(N,\chi_N)$ and $S^{\an}_k(N,\chi_N)$ denote the space of real-analytic functions on the upper-half plane with the same transformation properties
under $\Gamma_0(N)$ and having bounded growth (resp.\,rapid decay) at the cusps. On these spaces one may define the Shimura-Maass derivative operator
$$ \delta_k := \frac{1}{2\pi i}
\left( \frac{d}{dz}  + \frac{i k}{2y}\right): M^{\an}_k(N,\chi_N) \lra M^{\an}_{k+2}(N,\chi_N).$$



For every $k\geq 1$ such that $\chi(-1)=(-1)^k$, define the {\em non-holomorphic Eisenstein series} of weight $k$ and level $N$ attached to the character $\chi_N$ as the function on $\mathcal H \times \C$ given by the rule
\begin{equation}
	\tilde{E}_{k, \chiNN}(z,s) = \sum_{(m,n)\in \Z^2\setminus\{(0,0)\}} \frac{\chi_N^{-1}(n)}{(mNz+n)^k} \cdot \frac{y^s}{|mNz+n|^{2s}}.
\end{equation}

Although a priori this series only converges for $\Re(s) > 1 -k/2$, it can be extended to a meromorphic function in the variable $s$ on the whole complex plane $\C$. For $k>2$, or $k \geq 1$ but $\chi \ne 1$,  the series arising by setting $s=0$ is actually holomorphic in $z$ and gives rise to a modular form
\[
	\tilde{E}_{k, \chiNN}(z) := \tilde{E}_{k, \chiNN}(z,0) \in M_k(N, \chi).
\]

For any value of $s$, the series $\tilde{E}_{k, \chiNN}(z,s)$ belongs to $M_k^{\an}(N, \chi)$ and
one verifies that 
\[
	\delta_k \tilde{E}_{k,\chiNN}(z,s) = -\frac{s+k}{4\pi} \tilde{E}_{k+2,\chiNN}(z,s-1).
\]
Moreover, if we let $\delta_k^t = \delta_{k+2t-2} \cdots \delta_{k+2} \delta_k$ denote the $t$-fold iterate of the Shimura-Maass operator, then for all $t\leq (k-1)/2$ we have
\begin{equation}\label{ShimuraMaass}
	\tilde{E}_{k,\chiNN}(z, -t) = \frac{(k-2t-1)!}{(k-t-1)!}(-4\pi)^t \delta_{k-2t}^t \tilde{E}_{k-2t,\chiNN}(z).
\end{equation}

Define a normalization $E_{k,\chiNN}\in M_k(N, \chi)$ of the Eisenstein series as
\begin{equation} \label{trueEisenstein}
	E_{k,\chiNN}(z) = \frac{N^k (k-1)! }{2 (-2\pi i)^k \tau(\chi^{-1})}\cdot \tilde{E}_{k,\chiNN}(z)
\end{equation}
where 
$$
\tau(\chi) = \sum_{a=1}^{N_\chi} \chi(a) e^{\frac{2\pi a i}{N_\chi}}
$$
is the Gauss sum associated to the Dirichlet character $\chi$.


Let  $\sigma_{k-1,\chi}$ denote the function on the positive integers defined as $\sigma_{k-1,\chi}(n) := \sum_{d\mid n} \chi(d)d^{k-1}$. Then $E_{k,\chi}$ is a newform of level $N_\chi$ and its $q$-expansion is
\begin{equation}
	E_{k,\chi}(q) = \frac{L(\chi, 1-k)}{2}+\sum_{n=1}^\infty \sigma_{k-1, \chi}(n) q^n \in M_k(N_\chi,\chi), \qquad q=e^{2\pi i z}.
\end{equation}

When $N>N_\chi$, $E_{k,\chi_N}(q)$ is a $\Q_\chi$-linear combination of the modular forms $E_{k,\chi}(q^d)$ as $d$ ranges over the positive divisors of $N/N_\chi$ (cf.\,\cite[(3.3), (3.4)]{Sh76} for the precise expression). In particular $E_{k,\chi_N}$ is an eigenform with respect to all good Hecke operators $T_{\ell}$, $\ell\nmid N$ with the same eigenvalues of $E_{k,\chi}$.



\subsection{Classical Rankin's $L$-function}

Recall that the Petersson scalar product on the space of real-analytic modular forms $S_l^{\an}(N,\chi) \times M_l^{\an}(N,\chi)$ is given by:
\begin{equation} \label{Peterssondef}
	\langle f_1, f_2 \rangle_{l,N} := \int_{\Gamma_0(N) \backslash \mathcal H} y^l \overline{f_1(z)} f_2(z) \frac{dx dy}{y^2}.
\end{equation}
Let $$g_l = \sum_{n\geq 1} a_n(g_l)q^n \in S_l(N, \chi_g), \quad f_k=\sum_{n\geq 1} a_n(f_k)q^n \in M_k(N, \chi_f)$$ be two eigenforms of weights $l > k \geq 1$ and nebentype characters $\chi_g$ and $\chi_f$ respectively. We do not assume $g_l$ and $f_k$ to be newforms, but we do assume them to be eigenvectors for all good and bad Hecke operators.


Set $\chi := (\chi_g \chi_f)^{-1}$ and let $g_l^*  = \sum_{n\geq 1} \bar{a}_n(g_l)q^n \in S_l(N, \chi_g^{-1})$ denote the modular form whose fourier coefficients are the complex conjugates of those of $g_l$. 

For a rational prime $q$ we let $(\alpha_q(g_l),\beta_q(g_l))$ denote the pair of roots of the Hecke polynomial $X^2-a_q(g_l)X+\chi_{g,N}(q)q^{l-1}$, that we label in such a way that $\ord_q(\alpha_q(g_l)) \leq \ord_q(\beta_q(g_l))$. Note that $(\alpha_q(g_l),\beta_q(g_l)) = (a_q(g_l),0)$ when $q\mid N$. If the weight is $l=1$ and $q\nmid N$ then both $\alpha_q(g_l)$ and $\beta_q(g_l)$ are $q$-units; in that case we just choose an arbitrary ordering of this pair. Adopt similar notations for $f_k$. 

Define the {\em Rankin $L$-function} of the convolution of $g_l$ and $f_k$ as the Euler product 
\begin{equation}\label{EulHR}
L(g_l\otimes f_k,s) = \prod_q L^{(q)}(g_l\otimes f_k,s),
\end{equation}
where $q$ ranges over all prime numbers and
\begin{align*}
L^{(q)}(g_l\otimes f_k,s) =	 &(1-\alpha_q(g_l)\alpha_q(f_k) q^{-s})^{-1} (1-\alpha_q(g_l)\beta_q(f_k) q^{-s})^{-1} \\
						&\times (1-\beta_q(g_l)\alpha_q(f_k) q^{-s})^{-1} (1-\beta_q(g_l)\beta_q(f_k) q^{-s})^{-1}.
\end{align*}
\begin{proposition}[Shimura] For all $s\in \C$ with $\Re(s) >>0$ we have:
\begin{equation} \label{Shimura}
	L(g_l\otimes f_k,s) = \frac{1}{2} \frac{(4\pi)^s}{\Gamma(s)} \langle g_l^*(z), \tilde{E}_{l-k, \chiNN}(z,s-l+1) \cdot f_k(z) \rangle_{l,N}
\end{equation}
\end{proposition}

Choose integers $m,t$ such that $$l = k+m+2t \quad \mbox{ and set } \quad j = (l+k+m-2)/2 = l-t-1.$$ For $m\geq 1$ and $t\geq 0$, evaluating equation \eqref{Shimura} at $s=j$ and using equations \eqref{ShimuraMaass} and \eqref{trueEisenstein}  one finds that
\begin{equation} \label{rankinmethod}
	\fhr(l,k,m) \cdot L(g_l\otimes f_k, j) = \langle g_l^*(z), \delta_m^t E_{m,\chiNN}(z) \cdot f_k(z) \rangle_{l,N},
\end{equation}
where
\begin{equation}
	\fhr(l,k,m) = \frac{(-1)^t (m+t-1)! (j-1)! (iN)^m }{2^{l-1}(2\pi)^{l+m-1}\cdot \tau(\chi^{-1})}.
\end{equation}



\subsection{Critical values, algebraicity and the Hida-Rankin $p$-adic $L$-function}\label{secHida}

Since we are assuming $l>k\geq 1$, an integer $j$ is critical for $L(g_l\otimes f_k,s)$ if and only if $j\in [k, l-1]$. We shall restrict our attention to critical integers in the range
\[
	j \in \Bigl[\frac{l+k-1}{2}, l-1 \Bigr],
\]
and for a given such $j$ we set $$t:= l-j-1 \quad \mbox{and} \quad m:= l-k-2t.$$ From equation \eqref{rankinmethod} it follows that
\begin{equation} \label{rankinmethod2}
	\fhr(l,k,m) \cdot L(g_l\otimes f_k,j) = \langle g_l^*(z), \delta_m^t E_{m,\chiNN}(z) \cdot f_k(z) \rangle_{l,N}.
\end{equation}
Define the algebraic part of $L(g_l\otimes f_k, j)$ as in \cite[(9)]{BDR1}:
\begin{equation} \label{algebraicpart}
	L^{\alg}(g_l\otimes f_k,j) := \fhr(l,k,j)\frac{L(g_l\otimes f_k, j)}{\langle g_l^*, g_l^* \rangle_{l,N}} =\frac{\langle g_l^*(z), \delta_m^t E_{m,\chiNN}(z) \cdot f_k(z) \rangle_{l,N}}{\langle g_l^*, g_l^* \rangle_{l,N}}.
\end{equation}

Fix a prime $p\nmid N$ at which $g_l$ is ordinary and let $g_{l,\alpha} \in S_l(Np, \chi_g)$ denote the ordinary $p$-stabilisation of $g_l$ on which $U_p$ acts with eigenvalue $\alpha_p(g_l)$. 

 
Let $\hg$ be a Hida family of ordinary overconvergent modular forms of tame (but not necessarily primitive) level $N$, passing through $g_{l,\alpha}$. The Hida family is parametrized by a finite \'etale rigid-analytic cover $U_\hg$ of weight space $\cW$. By shrinking $U_\hg$ if necessary, we assume that $U_\hg(\Z_p)$ is fibered over a single residue class modulo $p-1$ of $\cW(\Z_p) = \Z_p^\times \simeq \Z/(p-1)\Z \times \Z_p$. By a slight abuse of notation which shall be harmless for our purposes, we identify throughout points in $U_\hg$ with their image in $\cW$ under the weight map.





%Note that if $l=1$ then $g_l$ is ordinary at all $p$; in such case we further assume that 
%$$
%\alpha_p(g_l) \ne \beta_p(g_l)
%$$


With these conventions, for every classical weight $l \in U_\hg\cap \Z^{\geq 2}$ we let
\begin{equation}\label{gl}
	g_l\in S_l(N,\chi_g)
\end{equation}
denote the classical cusp form whose ordinary $p$-stabilisation is the specialisation of $\hg$ at an arithmetic point in $U_{\hg}$ of weight $l$. 


%Consider then the region of interpolation:
%\[
%	U_{\text{int}} := \Bigl\{ (l,k,j) \in U_\hg\times U_\hf\times \Z \mid \quad h\in  U_\hg\cap \Z^{\geq 2}, k \in  U_\hf\cap \Z^{\geq 2}, \quad \frac{l+k-1}{2} \leq j \leq %l-1 \Bigr\}
%\]



Define $$\ehr(l,k,j) := \frac{\cE(g_l,f_k,j)}{\cE_1(g_l)\cE_0(g_l)}$$ where:
\begin{eqnarray*}
	\cE(g_l,f_k,j)&=&
	(1-\beta_p(g_l)\alpha_p(f_k) p^{t-l+1})(1-\beta_p(g_l)\beta_p(f_k) p^{t-l+1}) \\
	&\times& (1-\beta_p(g_l)\alpha_p(f_k) \chi(p) p^{t-l+1})(1-\beta_p(g_l)\beta_p(f_k) \chi(p) p^{t-l+1}),\\
	\cE_1(g_l)&=&1-\beta_p(g_l)^2 p^{-l}, \\
	\cE_0(g_l)&=&1-\beta_p(g_l)^2 p^{1-l}.
\end{eqnarray*}

In \cite[\S 7.4]{hida-book} Hida constructed a three-variable $p$-adic $L$-function interpolating central critical values of the Rankin $L$-function associated to the convolution of two Hida families of modular forms. For the purposes of this note it will suffice to retain the restriction of this $p$-adic $L$-function to the one-dimensional domain afforded by $U_\hg$. Here we will work with the notations and normalisations adopted in \cite{BDR1}. In order to introduce this $p$-adic $L$-function properly, we shall make use of the following operators on the space of overconvergent $p$-adic modular forms, that we introduce here by describing their action on $q$-expansions:

\begin{itemize}

\item Serre's derivative operator $d =q\cdot \frac{d}{dq}$, which may be regarded as the $p$-adic avatar of the Shimura-Maass operator invoked above.

\item The $U$ and $V$ operators acting on a modular form $\phi=\sum a_n q^n$ by the rules $$U(\phi) = \sum a_{pn}q^n \quad \mbox{and} \quad V(\phi) = \sum a_n q^{pn}.$$

\item Hida's ordinary idempotent  $e_{\ord} := \lim U_p^{n!}$.

\item $p$-depleting operator: $\phi^{[p]} := (1- UV)(\phi) = \sum_{p\nmid n} a_n q^n$. 

\end{itemize}

Let $\cE$ denote the Kuga-Sato variety fibered over $X_1(N)$ and let $\cE^{l-2}$ denote the fiber product of $l-2$ copies of $\cE$ over $X_1(N)$. The dimension of $\cE^{l-2}$ is $l-1$ and the middle de Rham cohomology group $H^{l-1}_{\dR}(\cE^{l-2}/\C_p)$ contains the canonical regular differential form $\omega_{g^*_l}$ associated to $g^*_l$. Let  
$$
\langle\ ,\ \rangle: H^{l-1}_{\dR}(\cE^{l-2}/\C_p) \times H^{l-1}_{\dR}(\cE^{l-2}/\C_p) \, \lra \, \C_p
$$ 
denote the non-degenerate Poincar\'e pairing on $H^{l-1}_{\dR}(\cE^{l-2}/\C_p)$.


The $g^*_l$-isotypic component of
$H^{l-1}_{\dR}(\cE^{l-2}/\C_p)$ is two-dimensional over $\C_p$ and it admits a one-dimensional 
{\em unit root subspace}, denoted $H^{l-1}_{\dR}(\cE^{l-2}/\C_p)^{\ur}$,
on which the Frobenius endomorphism acts as multiplication by a $p$-adic unit.
This unit root subspace is complementary to the line spanned by $\omega_{g^*_l}$. There is thus a single class
$\eta_{g^*_l}^{\ur} \in H^{l-1}_{\dR}(\cE^{l-2}/\C_p)^{\ur}$ satisfying
$$ \langle \omega_{g^*_l},\eta_{g^*_l}^{\ur} \rangle = 1.$$


%For all triples $(l,k,j)\in U_{\text{int}}$ we can define the values:
%\begin{equation} \label{definhida}
%	L_p^\hg(\hg,\hf)(l,k,j) := \frac{1}{\cE_0(g_l,f_k,j)}\langle \eta^{ur}_{g_l}, e_{\ord}(d^t E_{m,\chi}^{N, [p]} \cdot f_k) \rangle_{l,N}.
%\end{equation}

Let $\Lambda_g$ denote the algebra of Iwasawa functions on $U_{\hg}$ and let $\mathcal K_{\hg}$ denote the fraction field of $\Lambda_{\hg}$.  As shown in \cite{BDR1}, there exists a unique $p$-adic $L$-function $L_p(\hg,f) \in \mathcal K_{\hg}$ satisfying
\begin{equation} \label{definhida}
L_p(\hg,f)(l) = \frac{1}{\cE_0(g_l,f_k,j)}\langle \eta^{ur}_{g^*_l}, e_{\ord}(d^t E_{m,\chiNN}^{[p]} \cdot f_k) \rangle
\end{equation}
for all $l\in U_{\hg} \cap \Z_{\geq 2}$. Note that $d^t E_{m,\chiNN}^{[p]} \cdot f_k$ is an overconvergent modular form of weight $l\geq 2$ and hence its ordinary projection is classical by a celebrated theorem of Hida and Coleman. This way $e_{\ord}(d^t E_{m,\chiNN}^{[p]} \cdot f_k)$ gives rise to a regular differential form  in $H^{l-1}_{\dR}(\cE^{l-2}/\C_p)$, which may therefore be written as 
$$
e_{\ord}(d^t E_{m,\chiNN}^{[p]} \cdot f_k) = C(g_l,f_k)\cdot \omega_{g_l^*} + (\mbox{Other terms with respect to an orthogonal basis})
$$

In plane terms, the above formula might be read as
$$
L_p(\hg,f)(l) = \frac{C(g_l,f_k)}{\cE_0(g_l,f_k,j)}.
$$

The $p$-adic $L$-function $L_p(\hg,f_k)$ deserves its name because it obeys and it is characterized by the following interpolation formula that relates the values of $L_p(\hg,f_k)$ at integers $l>k$ to critical values of a Rankin $L$-function. In fact, it follows from \eqref{algebraicpart} and \eqref{definhida} that for every $l\in U_{\hg} \cap \Z_{l>k}$ we have
\begin{equation} \label{interpolhida}
	L_p(\hg,f_k)(l) = \ehr(l,k,j) \cdot L^{\alg}(g_l\otimes f_k,j) = \ehr(l,k,j) \fhr(l,k,j)\frac{L(g_l\otimes f_k, j)}{\langle g_l^*, g_l^* \rangle_{l,N}}. 
\end{equation}
We call this function the {\em Hida-Rankin $p$-adic $L$-function} associated to $\hg$ and $f_k$.

Note  that $l=1$ lies outside the above region of interpolation. Assume that there exists an eigenform $g_1\in M_1(N_g,\chi_g)$  such that the ordinary $p$-stabilisation $g_{1,\alpha}$ arises as the specialisation of $\hg$ at an arithmetic point of weight $1$ in $U_{\hg}$. Notice that this is not always the case, as a Hida family may in general specialize to non-classical overconvergent modular forms at points of weight one.

The following result provides a formula for the value of $L_p(\hg,f)$ at $l=1$ and was proved in \cite[\S 2]{DLR}. 
Assume that $k=2$ and $\chi_f=1$, so that $\chi=\chi_g^{-1}$, and set $f=f_2$. Recall the $p$-adic iterated integrals introduced in \eqref{iterated}.




\begin{proposition} \label{vitasalva}
 Let $h:=E_{1, \chiNN} \in M_1(N,\chi_N)$. Then $L_p(\hg,f)$ has no pole at $l=1$ and
\begin{equation} 
	L_p(\hg,f)(1) = \int_{\gamma_{g_1}} f \cdot h.
\end{equation}
\end{proposition}

\begin{proof}
Combine \cite[remark 4.5]{DR1}, \cite[proposition 4.6]{DR1} and \cite[proposition 2.6]{DLR}.
\end{proof}



%==========SECTION: p-ADIC L-FUNCTIONS==========
%\section{$p$-adic $L$-functions}

%As suggested in \cite{DLR}, the result follows by an appropriate factorisation of $p$-adic $L$-function. The interpolation formula \eqref{interpolhida} is our central %object. The Petersson product $\langle g^*, g^* \rangle_{l,N}$ can be related it to the value of the complex $L$-function $L(\psi^2, 0$). When $p$ splits in $K$ it %naturally leads to the Katz $p$-adic $L$-function, whose special values are related to the $p$-adic logarithm of elliptic units, as we desire. The complex function %$L(f, \psi, s)$ of \cite{BDP1} differs from $L(g\otimes f, s)$ only for a finite number of bad Euler factors as explained in loc. cit. An interesting $p$-adic analogue is %the Bertolini-Darmon-Prasanna $p$-adic $L$-function, whose special values are related to the elliptic logarithm of Heegner points.




%=====Subsection 1: Katz 2 variables p-adic $L$-function==========



%==========BDP $L$-functions==========
\subsection{Bertolini-Darmon-Prasanna's $p$-adic $L$-function}\label{BDPs}
%%%%%%%%%%%%%%%%%%%%%%%%

As in the introduction, let $E/\Q$ be an elliptic curve of conductor $N_E$ and let $f\in S_2(N_E)$ denote the eigenform associated to it by modularity. As in \S \ref{Katzs}, let also $K$ be an imaginary quadratic field of discriminant $-D_K \leq -7$ fulfilling the Heegner hypothesis. 

Let $\fc \subseteq \cO_K$ be an integral ideal and set $N=\mathrm{lcm}(N_E,D_K N_{K/\Q}(\fc))$. 
Let $\Sigma$ denote the set of Hecke characters of $K$ of conductor dividing $\fc$. For any Hecke character $\psi\in \Sigma$ of infinity type $(\ka_1,\ka_2)$, let $L(f,\psi,s)$ denote the $L$-function associated to the compatible system of Galois representations afforded by the tensor product  $\varrho_{f|G_K} \otimes \psi$ of the (restriction to $G_K$ of) the Galois representations attached to $f$ and the character $\psi$.

% $$L(f,\psi,s) := L(\pi_f \times \pi_\psi, s-\frac{k_1+k_2+1}{2})$$ denote the $L$-series associated with the product of the global automorphic representations %attached to the weight two cuspform $f$ and the Hecke character $\psi$, normalised conveniently. 

As usual, $L(f,\psi,s) = \prod_q L^{(q)}(q^{-s})$ is defined as a product of Euler factors ranging over the set of prime numbers. The Euler factors at the primes $q$ such that $q\nmid N$ are exactly the same as that of the Rankin L-series $L(\theta_\psi \otimes f,s)$ introduced above, but may differ at the primes $q$ such that $q\mid N$ (details can be found in \cite{Gro} for $f$ modular form of weight 2).
%: we refer to \cite[\S 15]{Jac} for more details.

Let $\Sigma_{f,K}\subset \Sigma$ be the subset of Hecke characters of trivial central character in $\Sigma$ for which $L(f, \psi^{-1},s)$ is self-dual and $s=0$ is its central critical point. This set is naturally the disjoint union of the three subsets 
$$
\Sigma^{(1)}_{f,K} = \{ \psi \in \Sigma_{f,K} \,\mbox{ of infinity type }\,(1,1)\},
$$
$$
\Sigma^{(2)}_{f,K} = \{ \psi \in \Sigma_{f,K} \,\mbox{ of infinity type }\,(2+\ka,-\ka), \ka\geq 0\}
$$
and
$$
\Sigma^{(2')}_{f,K} = \{ \psi \in \Sigma_{f,K} \,\mbox{ of infinity type }\,(-\ka,\ka+2), \ka\geq 0\}.
$$

Fix a prime $p=\wp \bar\wp$ that splits in $K$.  
Each of the three sets $\Sigma^{(1)}_{f,K}$, $\Sigma^{(2)}_{f,K}$ and $\Sigma^{(2')}_{f,K}$ are dense in the completion $\hat{\Sigma}_{f,K}$ of  $\Sigma_{f,K}$ with respect to the $p$-adic compact open topology as explained in \cite[\S 5.2]{BDP1}. As shown in \cite{BDP1}, there exists a unique $p$-adic analytic function
\begin{equation*}
L_p(f,K): \hat{\Sigma}_{f,K} \lra \C_p
\end{equation*}
interpolating the critical values $L(f,\psi^{-1},0)$ for $\psi \in \Sigma^{(2)}_{f,K}$, suitably normalized.

We refer to $L_p(f,K)$ as the Bertolini-Darmon-Prasanna $p$-adic Rankin $L$-function attached to the pair $(f,K)$. 

%Recall the unramified Hecke characters $\psi_k$ introduced in \eqref{hecke-chars2}. It will be sufficient for our purposes to consider the restriction of $L_p(f,K)$ to the line arising as the $p$-adic closure of the set of Hecke characters 
%\begin{equation*}
%\{ \Psi_k:=\psi^{-1}_{2+2k} \mathbf{N}_K^{2+k}  \}
%\end{equation*}
%of infinity type $(0,-2-2k)+(2+k,2+k) = (2+k,-k)$. For $k\geq 0$ the character $\Psi_k \in \Sigma^{(2)}_{f,K}(\mathfrak N)$ lies in the region of interpolation of $L_p(f,K)$ and according to \cite[\S 5.2]{BDP1} the interpolation formula reads

Consider a character $\Psi \in \Sigma^{(2)}_{f,K}$ of type $(2+\ka, -\ka)$, for $\ka \geq 0$. According to \cite[\S 5.2]{BDP1} the interpolation formula for $L_p(f,K)$ reads
\begin{equation}\label{int-BDP}
L_p(f,K)(\Psi) =\cE_c \cdot  \mathfrak{e}_{\mathrm{BDP}}(\Psi)   \cdot \mathfrak{f}_{\mathrm{BDP}}(\Psi)\cdot \frac{ \Omega_p^{4\ka+4}  }{\Omega^{4\ka+4}} \cdot L(f,\Psi^{-1},0),
\end{equation}
where

\begin{itemize}

\item
$\cE_c =  \prod_{q\mid c} \frac{q-\chi_K(q)}{q-1}$, 	$\qquad \mathfrak{e}_{\mathrm{BDP}}(\Psi) = (1-a_p(f) \Psi^{-1}(\bar\wp) + p\Psi^{-1}(\bar\wp) ^2 )^2$,
\item
	$\mathfrak{f}_{\mathrm{BDP}}(\Psi)  = \left(\frac{2\pi}{\sqrt{D_K}}\right)^{2\ka+1}  \ka!(\ka+1)!  \cdot 2^{\sharp q\mid (D_K,N_E)} \cdot \omega(f,\Psi)^{-1}$ 
\end{itemize}
with $\omega(f, \Psi)$ as defined in \cite[(5.1.11)]{BDP1}. 

If $\psi$ is a finite order anticyclotomic character of conductor $c\mid \mathfrak c$, then $\psi \mathbf{N}_K$ lies outside the region of interpolation and the main theorem of \cite{BDP2} asserts that
\begin{equation}\label{thm:bdp-gz}
L_p(f,K)(\psi^{-1} \mathbf{N}_K) = \mathfrak f_p(f,\psi) \times \log_{\omega_E}(P_{\psi})^2
\end{equation}
where $\mathfrak f_p(f,\psi) = (1-\psi(\bar\wp) p^{-1} a_p(f) + \psi^2(\bar\wp) p^{-1})^2$.












%==========MAIN THEOREM==========
\section{Proof of the main theorem}

%We now proceed with the proof of Theorem \ref{main-theta}.We keep the notations and assumptions as in the introduction and the previous sections.
Recall the three eigenforms that have been fixed at the outset:
\[
f\in S_2(N_E), \quad	 g=\theta_\psi \in M_1(D_Kc^2, \chi_K)_{\Q_{\psi}}, \quad h=\mathrm{E}_{1,\chi_K} \in M_1(D_K,\chi_K).
\]
%Since $\psi$ is anticyclotomic, the modular form $g$ is an Eisenstein series if and only if $\psi^2 = %1$. In this setting we have that:
%\[
%	\psi= \psi_1 = \psi_2, \qquad V:=V_1=V_2, \qquad L(E, \varrho_{gh}, s) = L(g \otimes f, s)^2,
%\]
%hence we can apply the Rankin-Selberg method to $L(g\otimes f, s)$ in order to get the result.
%\subsection{The choice of the test vectors}  In order to apply Rankin's method we need to raise the %levels of $f$ and $g$ to a common level $N$. We define $N:=\lcm(N_E,Dc^2)$ and 
Choose and fix modular forms $\testf \in S_2(N)[f]$ and $\testg \in M_1(N,\chi_K)[g]$ that are eigenforms for all good and bad Hecke operators. For the sake of concreteness, we may write
\[
\testf(z) = \sum_{d \mid \frac{N}{N_E}} \mu_d(f) f(d z), \quad	\testg(z) = \sum_{d \mid \frac{N}{Dc^2}} \mu_d(g) g(d z)
\]
where $\mu_d(f)$, $\mu_d(g)$ belong to the number field $\Q_{\psi}(f_N)$ introduced in the paragraph preceding Theorem \ref{main-theta}.

%where for each $d$ the coefficient $\mu_d(f)$ is either $\alpha_\lambda(g)$ or $\beta_\lambda(g)$. %Up to a multiplicative factor we have $2^{\#\lambda\mid N/Dc^2}$ choices pour such a modular %form. Similarly, we have $2^{\#\lambda \mid N/N_E}$ choices for $\testf$ and the analogous explicit %expression. We define $\testh = E_{1, \chi_{_{K,N}}}  \in M_1(N, \chi_K)[h]$ which is the only possible choice, in light of Rankin's method. 

%\begin{remark}
%	Notice that $\alpha_p(\testg) = \alpha_p(g)$ and $\beta_p(\testg)=\beta_p(g)$ since we are %assuming that $p\nmid N$. This holds similarly for $\testf$ and $\testh$.
%\end{remark}



%\subsection{Hida family for $g$ and interpolation on the line $(2l+3,2,l+2)$} 

Fix a prime $p\nmid N$ that splits in $K$ and choose a root $\alpha$ of $T^2-a_p(g)T+1$. As in the introduction let $\testg_\alpha\in M_1(Np,\chi_K)$ denote the $p$-stabilisation of $\testg$ on which $U_p$ acts with eigenvalue $\alpha$.

There exists a unique $p$-adic Hida family $\hg$ of theta series of tame level $D_Kc^2$ and tame character $\chi_K$ passing through $g_\alpha$. As in \S \ref{secHida} and \eqref{gl}, for every classical weight $l \in U_\hg\cap \Z^{\geq 2}$ we let
\begin{equation}
	g_l\in S_l(D_Kc^2,\chi_K)
\end{equation}
denote the classical newform whose ordinary $p$-stabilisation is the specialisation of $\hg$ at an arithmetic point in $U_{\hg}$ of weight $l$. Notice that at $l=1$ the modular form $g_l$ is still classical, by assumption, but it might not be a cusp form. In this case we have $g_1 = g\in M_1(D_kc^2, \chi_K)$.

For every such $l$ we can also explicitly describe the Hecke character $\psi_{l-1}$ of conductor $c$ and infinity type $(0,l-1)$ such that $g_l=\theta_{\psi_{\ell-1}}$. We do the same construction done in \cite[p. 235-236]{hida-book} and \cite[\S 3]{DLR}, but we slightly change conventions. Pick a $p$-adic unitary character $\lambda$ of conductor $c\bar\wp$ and infinity type $(0,1)$. Define then $\psi_{\ell-1}(\mathfrak q) := \psi(\mathfrak q) \langle \lambda(\mathfrak q) \rangle ^{\ell-1}$, then define $\psi_{\ell-1}(\bar\wp) := p^{\ell-1}/\psi_{\ell-1}(\wp)$. At any prime $q=\mathfrak q \bar{\mathfrak q}$ which is splits in $K$ we have
\begin{equation} \label{useful}
	\alpha_q(g_l) = \psi_{l-1}(\mathfrak q), \qquad \beta_q(g_l) = \psi_{l-1}(\bar{\mathfrak q}).
\end{equation}
Our running hypothesis on $p$ and the Heegner assumption imply that this is the case for $q=p$ and for any of the primes dividing $N$ but not $D_K$. 

Together with $\hg$, it will also be useful to consider  the $\Lambda$-adic family of modular forms
 $$\htg(q) = \sum_{d \mid \frac{N}{Dc^2}} \mu_d(g) \hg(q^d)$$ 
arising from our choice of $\testg$. Note that $\htg$ specializes to $\testg_\alpha$ in weight one.

%Define
%\begin{equation}\label{hecke-chars}
%\psi^{(p)}_{k-1}=\psi \langle \lambda \rangle^{k-1}
%\end{equation}
%for some Hecke character $\lambda$ of infinity type $(1,0)$ and conductor $\wp$. Two  such Hecke %characters $\lambda$, $\lambda'$ differ by a finite type character and hence $\langle \lambda' %\rangle = \langle \lambda \rangle$, so the right-hand side of \eqref{hecke-chars} does not depend %on the choice of $\lambda$. Then $\psi_{k-1}$ is defined as
%\begin{equation}\label{hecke-chars2}
%\psi_{k-1}(\mathfrak q) = \begin{cases} \psi^{(p)}_{k-1}(\mathfrak q) & \mbox{ if } \mathfrak q \ne %\wp \\ p^{k-1}/\psi^{(p)}_{k-1}(\bar\wp) & \mbox{ if } \mathfrak q = \wp.
%\end{cases}
%\end{equation}


Let $U_{\hg}^\circ$ denote the subset of $U_{\hg}$ consisting of classical points of weights of the form $2l+3\equiv 1 \pmod{p-1}$ with $l\in \Z_{\geq 1}$. According to the conventions about Hida families adopted in \S \ref{secHida}, note that $U_{\hg}^\circ$ is dense in $U_{\hg}$.


Set $j = l+2$, $t=l$ and $m=1$. Then the interpolation formula of \eqref{interpolhida} at points in $U_{\hg}^\circ$ reads as follows:
\begin{equation} \label{1variable}
	L_p(\htg,\testf)(2l+3,l+2) = \ehr(l) \cdot \fhr(l) \cdot \frac{L(\testg_{2l+3}\otimes \testf, l+2)}{\langle \testg_{2l+2}^*, \testg_{2l+2}^* \rangle_{l,N}} 
\end{equation}
where $\ehr(l) = \cE(2l+3,2,l+2)/(\cE_1(2l+3) \cE_0(2l+3))$, with:
 \begin{eqnarray*}
 \cE(2l+3,2,l+2)&=&
 (1-\alpha_f \beta_{\theta_{2l+3}} p^{-(l+2)})^2
 (1- \beta_f \beta_{\theta_{2l+3}} p^{-(l+2)})^2,\\
 \cE_1(2l+3)&=&1-\beta_{\theta_{2l+3}}^2 p^{-2l-3}, \\
\cE_0(2l+3)&=&1-\beta_{\theta_{2l+3}}^2 p^{-2l-2},
 \end{eqnarray*}
and
\begin{equation}\label{foo-BDR}
\mathfrak{f}_{\mathrm{HR}}(l)  = \frac{(-1)^l l!(l+1)!\cdot i \cdot N}{ 2^{4l+5} \pi^{2l+3}\cdot \tau(\chi_K)} = (-1)^l  \cdot \frac{l!(l+1)!\cdot N}{ 2^{4l+5} \pi^{2l+3}\cdot \sqrt{D_K}}
\end{equation}  
Here the last equality holds because $\tau(\chi_K) = i \sqrt{D_K}$ (cfr. \cite{Roh}).

% primes dividing $N_E$ but not dividing $D$ are split in $K$ and the fact that $(N_E, c)=1$ implies that $N_0 = N_{E, 0} \cdot c_0$, so that we find $\chi(N_0/D) = \chi(c_0)$. Recall that, since $\cond(\chi_K)=D$, $N_0$ is defined as the largest number such that $N_0/D$ is square-free and $D\mid N_0\mid N$). Here it becomes clear why it is important to choose the conductor such that $(c,D)=1$.


We now need the following two basic lemmas.

\begin{lemma}\label{proposition1} There exists a meromorphic function $ \cEul_N(s)$ such that the following factorisation formula holds
\[
	L(\testg \otimes \testf, s) = \cEul_N(s) \cdot L(f, \psi, s)
\]
and $\cEul_N(1) \in \Q_\psi(f_N)^\times$.
\end{lemma}

\begin{proof}
	Since we choose $\testf$ and $\testg$ to be eigenforms for all Hecke operators, we can use Euler products to compare the two L-functions. Their Euler factors are equal outside primes $q\mid N$. Then the factor $\cEul_N(s)$ is the product of the bad Euler factors at $q\mid N$ which encode this discrepancy. If we use the definitions of equation \eqref{EulHR} and of \cite[equation (20.2)]{Gro} we find:
	\begin{equation} \label{grossplus}
		\cEul_N(s) = \frac{\prod_{q\mid (N_E,D_K)} (1+q^{-s}) \prod_{q||N_E, q\nmid D_K} (1-a_q(f)q^{-s})^2 \prod_{q\mid D_K, q \nmid N_E} (1-a_q(f)a_q(g)q^{-s}+q^{1-2s} )}{\prod_{q\mid N} (1-\alpha_q(\testf)\alpha_q(\testg)q^{-s})}
	\end{equation}
	so that $\cEul_N(1)$ is a finite product of non-zero terms which lies in $\Q_\psi(f_N)$.
\end{proof}


\begin{lemma} \label{prop1}
For $l=-1$ and $l\in U_{\hg}^\circ$, let $\Psi_l$ be the Hecke character  $\Psi_l = (\psi_{2l+2})^{-1} \mathbf{N}^{l+2}$ of conductor $c$ and infinity type $(l+2,-l)$.
Then there exists a number
$
 \cEul^{HR}_N(l) \in \Q_{\psi}(f_N)
$
such that the following equality of critical $L$-values hold:
\[
			L(\testg_{2l+3}\otimes \testf, l+2) =\cEul^{HR}_N(l)   \cdot L(f, \Psi_l^{-1}, 0).
\]
Moreover, for $l=-1$ we have  $\cEul^{\mathrm{HR}}_N(-1) \ne 0$.
\end{lemma}

\begin{proof} The $L$-functions $L(\testg_{2l+3}\otimes \testf,s)$ and $L(g_{2l+3}\otimes f, l+2)$ are defined by an Euler product with exactly the same local factors at all primes $q$ except possible for the primes $q\mid N$. From the definition of L-factors we have that $ \cEul^{\mathrm{HR}}_N(l)$ lies in $\Q_{\psi}(f_N)$ for all $l$.

For $l=-1$ we have $\cEul_N^{HR}(-1) = \cEul_N(1) \in \Q_{\psi}(f_N)^\times$ by Lemma \ref{proposition1}
\end{proof}

Secondly, we have the following classical formula for the Petersson product, due essentially to H. Petersson (cf.\,\cite[Theorem 5.1]{Hi81}, \cite[Satz 6]{Pe}): 

\begin{proposition}\label{prop2} Set the index $\Im(N) :=[\mathrm{SL}_2(\Z): \Gamma_0(N)] = \prod_{q^{n_q}\mid \mid N} q^{n_q-1}(q+1)$ and define
\[
\mathfrak{f}_{Pet}(l) := \frac{\Im(N)}{\Im(D_Kc^2)}\cdot \frac{(2l+2)!}{2^{4l+4} \pi^{2l+3}} \cdot \frac{h_c \cdot \sqrt{D_Kc^2}}{w_c}
\]
where $h_c$ and $w_c$ are the class number and the number of roots of unity of the order $\cO_c$ of conductor $c$. Then
\begin{equation}\label{sym-formula}
\langle g_{2l+3}^*,g_{2l+3}^*\rangle_N =   \mathfrak{f}_{Pet}(l) \cdot L(\psi^2_{2l+2},2l+3).
\end{equation}

\end{proposition}

\vspace{0.2cm}

%	\begin{proof}
%		It's a matter of factoring $D(\theta, \theta, s)$ in a product involving two Riemann zeta functions, the Dirichlet $L$-function associated with $\chi_K$ and the Hecke $L$-function associated with $\psi_{2k+2}^2$. Then use Shimura's formula for the residue at $s = 2k+3$ (\cite[pag. 786]{Sh76}, being careful of the different normalization of the Petersson product) and Dirichlet class number formula. Finally, we invoke the classic formula:
%		\[
%			h_c = \frac{c \cdot h_K}{[ \cO_K^\times : \cO_c^\times ]} \cdot \prod_{q\mid c}(1-\chi_K(q)q^{-1})
%		\]
%		to convert information in terms of $\cO_c$.
%	\end{proof}


Let us introduce now the Hecke  character  $\Phi_l = \psi_{2l+2}^{-2}\mathbf{N}^{2l+3}$ of infinity type $(2l+3, -2l-1)$, and note that $\Phi_l$ lies in the region of interpolation for the Katz $p$-adic $L$-function. Since $L(\Phi_l^{-1},s) = L(\psi^2_{2l+2},s+2l+3)$, it follows from \eqref{sym-formula} that
%Comparing the residue at $s=2l+3$ of the Rankin-Selberg convolution $L$-function $D(g,g,s) = \sum_{n\geq 1} a_n(g)^2 \cdot n^{-s}$ for $g_{2l+3}$ and %$\testg_{2l+3}$ one finds the following formula:
\begin{align*}
\langle \testg_{2l+3}^*, \testg_{2l+3}^* \rangle_{l,N} &= \cEul_N^{Pet}(l) \cdot \langle g_{2l+3}, g_{2l+3} \rangle_{l,N} \\
&= \cEul_N^{Pet}(l) \cdot \fpet (l)\cdot L(\Phi_l^{-1},0)
\end{align*}
where $\cEul_N^{Pet}(l) \in \Q_{\psi}$ is a non-zero number arising from the discrepancy at the primes $q\mid N$ of the local Hecke polynomials of $g_{2l+3}$ and $\testg_{2l+3}$. 

 
%$\cE_{Pet}(l) = \prod_{q \mid N/Dc^2}\frac{q-\chi_K(q)}{q+\chi_K(q)}$.

Set
\[
	\cEul_N(l) = \frac{\cEul_N^{\mathrm{HR}}(l)}{\cE_c \cdot \cEul_N^{Pet}(l)} \qquad \text{and} \qquad \mathfrak f_\infty(l) :=  \frac{\mathfrak{f}_{\mathrm{HR}}(l) \cdot \mathfrak{f}_{K}(\Phi_l)}{\mathfrak{f}_{\mathrm{BDP}}(\Psi_l) \cdot \mathfrak{f}_{Pet}(l)}
\]
and define the function
$$
\mathfrak f: U^\circ_{\hg}  \lra \C_p, \quad   \mathfrak f(l) := \cEul_N(l) \cdot \mathfrak f_\infty(l).
$$

\begin{theorem} \label{factor} The function $\mathfrak f$ interpolates to a $p$-adic analytic function on $U_{\hg}$ and the following factorisation of $p$-adic $L$-series holds:
\begin{equation}\label{main-fact}
L_p(\htg,\testf)(2l+3) \times  L_p(K)(\Phi_l) = \mathfrak{f}(l) \cdot L_p(f,K)(\Psi_l).
\end{equation}
\end{theorem}

\begin{proof}

%By equation \eqref{vitasalva}, the statement of our theorem is a formula for the value of $L_p^\theta(\utheta,F)$ at the point $(1,2,1)$ lying outside the region of interpolation. To find this value, we are going to relate the function $L_p^\theta(\utheta,F)$ to the other $p$-adic $L$-function that we have introduced above, looking at the common region of interpolation (here $\utheta$ is the Hida family associated with the Hecke character described in \eqref{hecke-chars2} but with $\psi_g$ substituted by $\psi$).
%Under the running assumption that $N = D_K$ is squarefree, we find that
%\begin{equation} \label{simplifiedsymformula}
%	\mathfrak{f}_{Pet}(k) =  \frac{h_K \cdot \sqrt{D_K}}{2^{4k+5} \pi^{2k+3}} \cdot (2k+2)!
%\end{equation}
In our setting, the Bertolini-Darmon-Prasanna interpolation formula reads as 
\begin{align} \label{1variableBDP}
	L(f,\Psi_l^{-1},0) &= \frac{1}{\ebdp(\Psi_l)   \cdot \fbdp(\Psi_l) \cdot  \cE_c}\cdot \frac{ \Omega^{4l+4}  }{\Omega_p^{4l+4}} \cdot  L_p(f,K)(\Psi_l), \\
	\mathfrak{e}_{\mathrm{BDP}}(\Psi_l) &= (1-a_p(f) \psi_{2l+2}(\bar\wp) p^{-2-l} + \psi_{2l+2}(\bar\wp) ^2 p^{-2l-3} )^2,  \notag \\
\mathfrak{f}_{\mathrm{BDP}}(\Psi_l)  &= \left(\frac{2\pi}{c \sqrt{ D_K}}\right)^{2l+1}  l!(l+1)!  \cdot 2^{\sharp q\mid (D_K,N_E)} \cdot\omega(f,\Psi_l)^{-1}, \notag \\
\cE_c &= \prod_{q\mid c}\frac{q-\chi_K(q)}{q-1} \notag .
\end{align}
After replacing $k_1=2l+3$ and $k_2=-2l-1$ in the Katz interpolation formula we find:
\begin{align}\label{1variableKatz}
L(\Phi_l^{-1},0) &=  \frac{1}{\mathfrak{e}_{K}(\Phi_l) \cdot \mathfrak{f}_{K}(\Phi_l)} \cdot \frac{\Omega^{4l+4}}{\Omega_p^{4l+4}} \cdot L_p(K)( \Phi_l), \\
\mathfrak{e}_{K}(\Phi_l) &= (1-\psi_{2l+2}^{-2}(\wp) p^{2l+2}) (1-\frac{\psi_{2l+2}^{2}(\bar\wp)}{p^{2l+3}}),  \notag \\
\mathfrak{f}_{K}(\Phi_l) &= \left( \frac{2 \pi}{\sqrt{D_K}}\right)^{2l+1}(2l+2)!  \notag 
\end{align}
Subsequently substitute equations \eqref{sym-formula}, \eqref{1variableBDP} and \eqref{1variableKatz} into equation \eqref{1variable}. Notice that by equation \eqref{useful} we have $\ehr(l) \ekatz(l) = \ebdp(l)$. An elementary manipulation immediately shows that the decomposition formula \eqref{main-fact} holds at all the points in $U_{\hg}^\circ$.

Since $U_{\hg}^\circ$ is dense in $U_{\hg}$, in order to complete the proof of the theorem it only remains to prove the claim that  $\mathfrak f$ extends to a $p$-adic analytic function on $U_{\hg}$. 

Recall that $\mathfrak f(l)$ was defined as the product of two functions $\cEul_N(l)$ and $\mathfrak f_\infty(l)$. The Euler factors encoded in $\cEul_N(l)$ are rational functions on powers of primes $q \mid N$ and hence $\cEul_N$ extends to a $p$-adic analytic function on $U_{\hg}$. 

As for the function $\mathfrak f_\infty(l)$, we first notice that from \cite[equation (5.1.11)]{BDP1} we can compute that $\omega(f,\Psi_l) = (-1/N)^{l+1} \cdot \psi_{2l+2}(\mathfrak N)^{-1}$. Hence $\omega(f,\Psi_l)$ also interpolates to $p$-adic analytic function on $U_{\hg}$. Combining the explicit recipes given in the text for $\mathfrak{f}_{\mathrm{HR}}(l)$, $\mathfrak{f}_{K}(\Phi_l)$, $\mathfrak{f}_{\mathrm{BDP}}(\Psi_l)$ and $\mathfrak{f}_{Pet}(l)$ one readily checks that
\begin{equation}\label{foo}
\mathfrak{f}_\infty(l)  =  - \frac{\Im(D_Kc^2)}{\Im(N)}\cdot \frac{N\cdot 2^{-\sharp {q\mid (D_K,N_E)}}}{h_c\cdot D_K}  \cdot \frac{\psi_{2l+2}(\mathfrak N)}{c^{-2l} \cdot N^{l+1}}.
\end{equation}
From this explicit description the claim follows.
\end{proof}



% \subsection{Main theorem} We prove here the main theorem and discuss the rationality of the fudge factor.

Thanks to the above result we can already prove Theorem \ref{main-theta}. We evaluate  \eqref{main-fact} at the point of weight one in $U_{\hg}$ arising when we set $l=-1$. In this case %$\Psi_{-1} = \psi^{-1} \mathbf{N}$ and $\Phi_{-1}= \psi^{-2}\mathbf{N}$, hence 
Theorem \ref{factor} asserts that
\begin{equation}\label{main-fact-one}
L_p(\htg,\testf)(1) \times  L_p(K)( \psi^{-2}\mathbf{N}) = \mathfrak{f}(-1) \cdot L_p(f,K)(\psi^{-1} \mathbf{N}).
\end{equation}

Since $L_p(K)((\psi)^{-2}\mathbf{N}_K) = L_p(K)(\psi^{-2})$ (c.f. \cite[p. 90-91]{Gr}), Proposition \ref{vitasalva} combined with equations \eqref{thm:bdp-gz} and \eqref{Katz-GZ} show that
\begin{equation}\label{main1}
\int_{\testgamma}\testf \cdot E_{1,\chi_{_{K,N}}} = \lambda(\testf, \testg) \cdot \frac{\log^2_{E,p}(P_\psi)}{\log_p(u_g)}.
\end{equation}
where 	
$$
\lambda(\testf, \testg) = \cEul_N(-1) \cdot  \mathfrak f_\infty(-1) \cdot \frac{\mathfrak f_p(f,\psi)}{\mathfrak f_p(\psi)}.
$$

From equation \eqref{foo} we hence derive that
$$
\mathfrak{f}_{\infty}(-1) = -\frac{\Im(D_Kc^2)}{\Im(N)} \cdot \frac{ N}{D_Kc^2 \cdot h_c} \cdot 2^{-\sharp {q\mid (D_K,N_E)}}\cdot \psi(\mathfrak N) \in \Q_\psi(f_N)^\times.
$$
By Lemma \ref{prop1}, $\lambda(\testf, \testg) \in \Q_{\psi}(f_N)^\times$ hence the first statement of Theorem \ref{main-theta} follows.

For the second part of our main theorem, assume that $N=D_K=N_E$ and $c=1$, and put $\testf=f$, $\testg=g$ and $\testh = h = E_{1,\chi_{_{K}}}$. In this setting $\cEul_N(-1) = 1$ and $$\lambda(f,g) =  \mathfrak f_{\infty}(-1) \cdot \frac{ \mathfrak f_p(f,\psi)}{\mathfrak f_p(\psi)}.$$ Moreover it follows from genus theory that $g_K := [\Cl_K: \Cl_K^2] = 2^{\#q\mid D_K -1}$ so that $\mathfrak f_\infty(-1) = \frac{-1}{2h_Kg_K}$.
The proof follows after combining that with formulae \eqref{Katz-GZ} and \eqref{thm:bdp-gz}.  


%Theorem \ref{main-theta} is  a prototypical example of a situation where the the numerical evaluation of $p$-adic  $L$-series attached to an elliptic curve $E$  can be used to recover points of infinite order on  $E$. The first example  of a formula of this type was discovered by Karl Rubin in \cite{rubin}, and concerns  the value of  the Katz two-variable $p$-adic $L$-function attached to a CM elliptic curve $A$ at a point that lies {\em outside} the range of classical interpolation defining it. Rubin's formula  expresses  this special value in terms of the (square of) the $p$-adic formal group  logarithm of a rational point on $A$.  The reader should consult \cite{rubin} or \cite{rubin-bu} for an exposition of Rubin's formula, and \cite{BDP2} for an explanation of how this formula can be deduced from \eqref{thm:bdp-gz}.


%\begin{remark}
%	Notice that for any choice of the test element $\testf$ we have our result but with different Euler factors appearing. If we sum over all possible choices of %$\testf$ we find:
%	\[
%		\testf' (z) = f(z) -\frac{1}{2^{\#q\mid N/N_E}} \sum_{q\mid N/N_E} a_q(f) \cdot f(qz).
	%\]
%	Using the linearity of the integral one can show that:
	%\[
	%	\int_{\testgamma}\testf' \cdot \testh = \sum_{\testf} \int_{\testgamma}\testf \cdot \testh = \Bigl( \sum_{\testf} \lambda(\testf, \testg, \psi) \Bigr) \cdot %\frac{\log^2_{E,p}(P_\psi)}{\log_p(u_g)}=  \lambda(\testf', \testg, \psi) \cdot \frac{\log^2_{E,p}(P_\psi)}{\log_p(u_g)}.
	%\]
	%It is then an easy Galois-theoretical exercise to prove that: $\lambda(\testf', \testg, \psi) \in \Q_{\psi}$.
%\end{remark}






\end{document}




\section{The Petersson scalar product and Hecke L-series}\label{PIP}

We now prove a formula for the Petersson scalar product associated to theta series following the method of proof of the related result in \cite[Theorem 5.1]{Hi81}. The argument is a straightforward derivation from a formula of Petersson \cite[Satz 6]{Pe}, but care must be taken over the various normalisations of the Petersson scalar product which occur in the literature.

\begin{proposition}
	Consider the cuspidal theta series $\theta^{*}_{2k+3}$ of integer weight $2k + 3 \geq 3$, character $\chi_K$, squarefree level $D=D_K$, associated to the Hecke character $\psi_{2k+2}$. Then:
	\begin{equation}\label{sym-formula}
	\langle \theta_{2k+3}^*,\theta_{2k+3}^*\rangle_N =   \mathfrak{f}_{Pet}(k) L(\psi^2_{2k+2},2k+3),
\end{equation}
	where, setting $\psi(N)=[\mathrm{SL}_2(\Z): \Gamma_0(N)] = \prod_{q^{n_q}\mid \mid N} q^{n_q-1}(q+1)$, we have:
	\[
		\mathfrak{f}_{Pet}(k) := \psi(N)\frac{(2k+2)!}{2^{4k+4} \pi^{2k+3}} \cdot \frac{\sqrt{D_K}\cdot h_K}{w_K \prod_{q\mid D_K} (q+1) }
	\] 
\end{proposition}

\begin{proof}
To ease notation write $\theta :=\theta^*_{2k+3}$ and $\psi := \psi_{2k+2}$. We show that
\begin{equation}\label{symD-formula}
\langle \theta,\theta \rangle_D =  \psi(D_K) \frac{(2k+2)! }{2^{4k+4} \pi^{2k+3}} \cdot \frac{\sqrt{D_K}\cdot  h_K}{w_K \prod_{p\mid D} (p+1) } \cdot L(\psi^2,2k+3)
\end{equation}
where $\psi(D)=[\mathrm{SL}_2(\Z): \Gamma_0(D)]$. Equation (\ref{sym-formula}) then follows by observing that $D$ divides $N$ and $\langle \cdot, \cdot \rangle_N = 
[\Gamma_0(D):\Gamma_0(N)]\langle \cdot, \cdot \rangle_D$. %(Our formula and proof are also valid for {\it cuspidal} theta series in weight $2k + 3 = 1$, i.e., when $\psi^2 \ne 1$ with $\psi$ of type $(0,0)$, although we do not need this case.)

Recall from \eqref{firstpetersson} that our normalization for the Petersson scalar product is, in this case:
$$
\langle \theta,\theta \rangle_D := \int_{\Gamma_0(D)\backslash \mathcal H} \theta(z)\overline{\theta(z)} y^{2k+1}dxdy.
$$


Write $L(\theta,s) = \sum_{n \geq 1} a_n(\theta) n^{-s} = \prod_p \left[(1 - \alpha_p p^{-s})(1 - \beta_p p^{-s})\right]^{-1}$, where $\alpha_p \beta_p = \chi_K(p)p^{2k+2}$ and $\alpha_p + \beta_p = a_p(\theta)$ whenever $p \nmid D$, and where $\beta_p = 0$ whenever $p \mid D$. We define the Rankin-Selberg convolution $L$-series 
\[
	D(\theta,\theta,s) := \sum_{n \geq 1} a_n(\theta)^2 n^{-s}
\]
and note that it has a local expansion \cite[(5.6$_{\rm b}$)]{Hi81}
\[ D(\theta,\theta,s) = \prod_p \left[ (1 - \alpha_p^2 \beta_p^2 p^{-2s})^{-1} (1 - \alpha_p^2 p^{-s})
(1- \alpha_p \beta_p p^{-s})^2 (1 - \beta_p^2 p^{-s}) \right]^{-1}.\]
By comparing local expansions it follows that, c.f. \cite[(0.4)]{Sh75}
\begin{equation}\label{Eqn-Dfact}
	\zeta_D(2s - 2(2k+3) + 2) D(\theta,\theta,s) = L(\sym^2 \theta, s) L(\chi_K,s - (2k+3) + 1)
\end{equation}
where $L(\sym^2 \theta, s) = \prod_p [(1 - \alpha_p^2 p^{-s})(1 - \alpha_p \beta_p p^{-s})(1 - \beta_p^2 p^{-s})]^{-1}$
is the symmetric square L-series, $\zeta_D(s)$ the Riemann zeta function with local factors at
primes $p|D$ removed, and $L(\chi_K,s)$ the Dirichlet $L$-series. By comparing local expansions once again one checks the factorisation
\begin{equation}\label{Eqn-sym2fac}
	L(\sym^2 \theta, s) = L(\psi^2,s)\zeta_D(s - (2k+3) + 1)
\end{equation}
where $L(\psi^2,s)$ is the L-series of the Hecke character $\psi^2$. (Note that the local factor of $L(\psi^2,s)$ at split primes $p$ of $K$ is $(1 - \alpha_p^2 p^{-s})(1 - \beta_p^2 p^{-s})$ but at inert primes it is $1 - p^{4k+2} p^{-2s}$ --- this is why we have the Riemann zeta function rather than a Dirichlet L-series in the equation). Now, the Petersson formula states that 
\begin{equation}\label{Eqn-Pet}
	\Res_{s = 2k+3} D(\theta,\overline{\theta},s) = (4 \pi)^{2k+3} \Gamma(2k+3)^{-1} \langle \theta,\theta \rangle_{Sh}.
\end{equation}
We give here the elegant formulation of Shimura \cite[(2.5)]{Sh76} (with a minor misprint
corrected by Hida \cite[Page 247]{Hi81}), using a different normalisation $\langle  \cdot, \cdot \rangle_{Sh}$
of the inner product. Looking at \cite[Page 786]{Sh76} one sees that $\langle  \theta , \theta \rangle_D = (\pi/3) \psi(D) \langle  \theta, \theta \rangle_{Sh}$ with $\psi(D) = [\mathrm{SL}_2(\Z): \Gamma_0(D)]$. Furthermore, since theta series have totally real coefficients we have $\overline{\theta} = \theta$. Then (\ref{Eqn-Pet}) can be rewritten as
\begin{equation}\label{Eqn-Res1}
\Res_{s = 2k+3} D(\theta,\theta,s) = \frac{3}{\pi \psi(D)} (4 \pi)^{2k+3} \Gamma(2k+3)^{-1} \langle \theta,\theta \rangle_D.
\end{equation}
Putting together (\ref{Eqn-Dfact}) and (\ref{Eqn-sym2fac}) we get
\begin{equation}\label{Eqn-Dtts}
	D(\theta,\theta,s) = \frac{\zeta_D(s - (2k+3) + 1)}{\zeta_D(2s - 2(2k+3) + 2)} L(\psi^2,s) L(\chi_K,s - (2k+3) + 1).
\end{equation}
The right-hand side of (\ref{Eqn-Dtts}) has a simple pole at $s = 2k+3$ coming from the simple pole of the Riemann
zeta function at $s = 1$. Thus from (\ref{Eqn-Dtts}) we see that
\begin{equation}\label{Eqn-Res2}
\Res_{s = 2k+3} D(\theta,\theta,s) = \left[ \frac{\pi^2}{6} \prod_{p|D_K} \left(1 - \frac{1}{p^2}\right) \right]^{-1} L(\psi^2,2k+3)
\prod_{p|D_K} \left(1 - \frac{1}{p} \right)
L(\chi_K,1).
\end{equation}
Our formula (\ref{symD-formula}) now follows immediately by comparing (\ref{Eqn-Res1}) and (\ref{Eqn-Res2}) and
invoking Dirichlet's class number formula $L(\chi_K,1) = 2 \pi h_K / w_K \sqrt{D_K}$.

\end{proof}





